COURT FILE NUMBER	2201-09319 Clerk's Stamp
COURT	COURT OF KING'S BENCH OF ALBERTA
JUDICIAL CENTRE	CALGARY
PLAINTIFF	ARNAKI LTD.
DEFENDANT	SOLVAQUA INC.
DOCUMENT	AFFIDAVIT
ADDRESS FOR SERVICE AND CONTACT INFORMATION OF PARTY FILING THIS DOCUMENT	Cassels Brock & Blackwell LLP Bankers Hall West 3810, 888 3rd St SW Calgary, AB T2P 5C5 P: 403 351 2920 E: joliver@cassels.com / dmarechal@cassels.com Attention: Jeffrey Oliver / Danielle Marechal
AFFIDAVIT OF:	TIFFANY NESS
SWORN ON:	October 19 2022
, Tiffany Ness, of the City of C	algary, in the Province of Alberta, SWEAR AND SAY THAT:

- 1. I am employed as a Legal Assistant with the firm of Cassels Brock & Blackwell LLP ("**Cassels**") and as such, have personal knowledge of the matters hereinafter deposed to, except where stated to be based on information and belief, in which case I believe the same to be true.
- 2. Terms not otherwise defined herein shall have the meaning ascribed to them in First Report of the Receiver dated October 11, 2022 (the "**Report**") and filed in the within proceedings.
- Attached hereto and marked as Exhibit "A" is a revised, final version of the Stalking Horse Agreement dated October 17, 2022 and executed by the Stalking Horse Purchaser (the "Final Stalking Horse Agreement").
- 4. Attached hereto and marked as **Exhibit "B"** is a blackline of the Final Stalking Horse Agreement to the version of the Stalking Horse Agreement attached as Schedule 2 to the Report.

SWORN BEFORE ME at Calgary, Alberta, this 19th day of October, 2022

(Commissioner for Oaths/Notary Public in and for Alberta)

Tiffan

JODIE Kosman A Commissioner for Oaths In and for Alberta My Commission Expires JUNE 3, 2024 THIS IS EXHIBIT "A' TO THE AFFIDAVIT OF TIFFANY NESS

SWORN ON THIS 19TH DAY OF OCTOBER 2022

<

A Commissioner for Oaths in and for Alberta

JODIE Kosman A Commissioner for Oaths In and for Alberta My Commission Expires

ASSET PURCHASE AGREEMENT

MNP LTD., solely in its capacity as receiver and manager of SOLVAQUA INC. and not in its personal capacity

as Seller

- and -

2464525 ALBERTA LTD.

as Buyer

October 17, 2022

00609155-1 LEGAL*56914030.9

PURCHASE AGREEMENT

THIS AGREEMENT is made as of October 17, 2022

BETWEEN:

MNP LTD. ("MNP"), solely in its capacity as receiver and manager of the assets, undertakings and properties of SOLVAQUA INC. ("SolvAQUA" or the "Debtor"), a corporation formed under the laws of the Province of Alberta, and not in its personal capacity

(the "Seller" or the "Receiver")

- and -

2464525 ALBERTA LTD., a corporation incorporated under the laws of the Province of Alberta

(and any of its nominees, designees or assignees hereunder, as applicable, the "Buyer")

RECITALS:

- A. The Debtor carries on the business of providing proprietary wastewater management solutions, based on nanopolymerization technology, and implementing that technology in foreign jurisdictions (together with any other business in which the Debtor is engaged on the date hereof, the "Business").
- B. Pursuant to the order of the Honourable Justice Neufeld of the Alberta Court of Queen's Bench (the "Court"), pronounced August 19, 2022 (the "Receivership Order") in Court File No. 2201-09319 (the "Receivership Proceedings"), MNP was appointed as the receiver and manager of all of the assets, undertakings and properties of the Debtor (the "Property").
- C. Pursuant to the Receivership Order, the Receiver is authorized to:
 - (a) market any or all of the Property and negotiate such terms and conditions of sale as the Receiver in its discretion may deem appropriate; and
 - (b) sell, convey, transfer, lease or assign the Property or any part or parts thereof out of the ordinary course of business with the approval of the Court in respect of any transaction in which the purchase price exceeds \$50,000.
- D. The Receiver: (i) wishes to sell, and the Buyer wishes to purchase (and has agreed to act as a "stalking horse bidder" pursuant to the SISP (as defined herein) substantially all the assets and liabilities of the Business; and (ii) intends to seek a Court order approving this Agreement and the SISP.
- E. The Receiver has determined that it is in the best interests of the Debtor's stakeholders to enter into this Agreement and to consummate the transactions contemplated herein on the terms set forth herein.

NOW THEREFORE in consideration of the mutual covenants and agreements contained in this Agreement and other good and valuable consideration (the receipt and sufficiency of which are acknowledged), the Parties agree as follows:

ARTICLE 1 INTERPRETATION

1.1 Definitions

In this Agreement,

- (a) "Adjustment Date" has the meaning given to such term in Section 3.6;
- (b) "affiliate" of any Person means, at the time such determination is being made, any other Person controlling, controlled by or under common control with such first Person, in each case, whether directly or indirectly through one or more intermediaries, and "control" and any derivation thereof means the control by one Person of another Person in accordance with the following: a Person ("A") controls another Person ("B") where A has the power to determine the management and policies of B by contract or status (for example the status of A being the general partner of B) or by virtue of beneficial ownership of a majority of the voting interests in B; and for certainty and without limitation, if A owns shares to which are attached more than 50% of the votes permitted to be cast in the election of directors (or other Persons performing a similar role) of B, then A controls B for this purpose;
- (c) "Agreement" means this Asset Purchase Agreement and all attached Schedules, in each case as the same may be supplemented, amended, restated or replaced from time to time, and the expressions "hereof", "herein", "hereto", "hereunder", "hereby" and similar expressions refer to this Agreement and all attached Schedules and unless otherwise indicated, references to Articles, Sections and Schedules are to Articles, Sections and Schedules in this Agreement;
- (d) "Applicable Law" means any domestic or foreign statute, law (including the common law), ordinance, rule, regulation, restriction, by-law (zoning or otherwise), order, or any consent, exemption, approval or license of any Governmental Authority, that applies in whole or in part to the Transaction, the Seller, the Debtors, the Buyer, the Business or any of the Purchased Assets;
- (e) "Arnaki" means Arnaki Ltd., a corporation incorporated under the laws of the British Virgin Islands;
- (f) "Arnaki Loan" means the loans advanced to the Debtor pursuant to the Arnaki Loan Agreements;
- (g) "Arnaki Loan Agreements" means the following loan agreements (as amended and restated) among Arnaki and the Debtor:
 - (i) Ioan agreement dated August 31, 2020;
 - (ii) Ioan agreement dated January 4, 2021; and
 - (iii) Ioan agreement dated February 1, 2021;
- (h) "Arnaki Loan Contracts" means the following:
 - (i) Arnaki Loan Agreements;
 - (ii) Arnaki Security;

- (iii) Assignment of Receivables Agreement dated February 1, 2021 between the Debtor and Arnaki; and
- (iv) Extension Agreement dated April 1, 2021 between the Debtor and Arnaki;
- (i) "Arnaki Security" means the following security agreements among Arnaki and the Debtor entered into pursuant to the Arnaki Credit Agreements:
 - (i) general security agreement dated August 31, 2020;
 - (ii) general security agreement dated January 4, 2021; and
 - (iii) general security agreement dated February 1, 2021;
- (j) "Assumed Liabilities" means those Liabilities incurred after Closing and relating to, or arising in connection with, the Purchased Assets, provided that such Liabilities continue to attach to the Purchased Assets following the granting of the Vesting Order;
- (k) "Business" has the meaning given to such term in Recital A;
- "Business Day" means any day, other than a Saturday or Sunday, on which the principal commercial banks in Calgary are open for commercial banking business during normal banking hours;
- (m) "Buyer" has the meaning given to such term in the preamble to this Agreement, or its nominee;
- (n) "Cash Component" means an amount to be determined by the Parties, acting reasonably, provided that such amount is equal to or greater than the sum of the Priority Payables and the Holdback Amount and may include the Vivakor Equipment Purchase Price if it is determined that (i) the Vivakor Equipment is owned by a Person other than SolvAQUA, and (ii) the Buyer elects to include the Vivakor Equipment in the Purchased Assets;
- "Claim" means any right, claim, cause of action or complaint of any Person that may be (0)asserted or made in whole or in part against the Debtor or Seller, any of their respective affiliates and their respective Representatives, whether or not asserted or made in connection with any indebtedness, liability or obligation of any kind whatsoever, and any interest accrued thereon or costs payable in respect thereof, including by reason of the commission of a tort (intentional or unintentional), by reason of any breach of contract or other agreement (oral or written), by reason of any breach of duty (including, without limitation, any legal, statutory, equitable or fiduciary duty) or by reason of any right of ownership of or title to property or assets or right to a trust or deemed trust (statutory, express, implied, resulting, constructive or otherwise), and whether or not any indebtedness, liability or obligation is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured, unsecured, present or future, known or unknown, by guarantee, surety or otherwise, and whether or not any right, claim, cause of action or complaint is executory or anticipatory in nature:
- (p) "Closing" means the completion of the Transaction at the Closing Time;
- (q) "Closing Date" means December 21, 2022, or such other date as the Parties may agree, acting reasonably;

- (r) "Closing Documents" means all contracts, agreements and instruments required by this Agreement to be delivered at or before the Closing;
- (s) "Closing Time" means 10:00 a.m. (Calgary time) on the Closing Date or such other time on the Closing Date as the Parties agree in writing that the Closing Time shall take place;
- (t) "Confidential Information" means non-public, confidential, personal or proprietary information which is furnished to a Party by the other Party, including, without limitation, information about identifiable individuals, any information relating to a Party and its affiliates or any customer or supplier of a Party; provided that "Confidential Information" does not include information that:
 - is or becomes generally available to the public other than as a result of disclosure in breach of this Agreement;
 - (ii) is received by a Party from a third party that obtained it lawfully and was under no duty of confidentiality;
 - (iii) was lawfully in a Party's possession prior to disclosure thereof by the other Party; or
 - (iv) was independently developed by a Party without use of, or reference to, the other Party's Confidential Information;
- "Contracts" means contracts, licences, permits, leases, agreements, commitments, entitlements or engagements to which the Debtor is party or by which the Debtor is bound;
- (v) "Court" has the meaning given to such term in Recital B;
- (w) "Court Approval" means the pronouncement of the SISP Order and the Vesting Order by the Court;
- (x) "CRA Priority Payments" means those deemed trust amounts validly owing by the Debtor to Canada Revenue Agency under the *Income Tax Act* (Canada) or the *Excise Tax Act* (Canada), or other similar provincial legislation as of the Closing Date that are in priority to the Arnaki Security;
- (y) "Credit Bid Component" has the meaning given to such term in Section 3.1;
- (z) "Debtor" has the meaning given to such term in the preamble to this Agreement;
- (aa) "Deposit" means the amount of \$100,000, comprising part of the Purchase Price;
- (bb) "Employee Priority Payments" means all amounts payable by the Debtor's estate to current or former employees of the Debtor under Sections 81.4 and 81.6 of the Bankruptcy and Insolvency Act (Canada) that rank in priority to the Arnaki Security;
- (cc) "Encumbrances" means all mortgages, pledges, charges, liens, hypothecs, hypothecations, debentures, trust deeds, Claims, assignments by way of security or otherwise, security interests, Priority Charges, conditional sales contracts or other title retention agreements, security created under the Bank Act (Canada), rights of first refusal, reservations of ownership or similar interests or instruments charging or creating a security interest in the Purchased Assets or any part thereof or interest therein, and any agreements, leases, licenses, occupancy agreements, options, easements, rights of way, restrictions, executions, or other encumbrances (including notices or other registrations in

respect of any of the foregoing) affecting title to the Purchased Assets or any part thereof or interest therein;

- (dd) "Equipment Owner" means the legal owner of the Vivakor Equipment;
- (ee) "Excluded Assets" means the Purchase Price;
- (ff) "Expense Reimbursement and Transaction Fee" has the meaning given to such term in Section 9.3(a);
- (gg) "Final" with respect to any order of any court of competent jurisdiction, means that leave to appeal or reconsideration shall not have been sought in respect of such order and that such order shall not have been stayed, appealed, varied (except with the consent of the Buyer and Seller) or vacated, and all time periods within which leave to appeal and reconsideration could at law be sought shall have expired or been waived and all time periods within which such order could at law be appealed shall have expired;
- (hh) "GST" means the sales tax payable under the GST Legislation;
- "GST Legislation" means Part IX of the Excise Tax Act (Canada);
- (jj) "Governmental Authority" means any government, regulatory authority, governmental department, agency, commission, bureau, court, judicial body, arbitral body or other law, rule or regulation-making entity:
 - having jurisdiction over the Debtor, the Buyer or the Purchased Assets on behalf of any country, province, state, locality or other geographical or political subdivision thereof; or
 - exercising or entitled to exercise any administrative, judicial, legislative, regulatory or taxing authority or power;
- (kk) "Governmental Authorizations" means the permits, licences, approvals and authorizations, orders, certificates, consents, directives, notices, licences, permits, variances, registrations or other rights issued to or held or required by the Seller and/or the Debtors relating to the Business or any of the Purchased Assets by or from any Governmental Authority;
- (II) "Holdback Amount" means the amount of \$75,000 paid by the Buyer to the Seller to account for any current or future Priority Payables, the amount of which Priority Payables cannot be definitively ascertained as at the Closing Date;
- (mm) "including" and "includes" shall be interpreted on an inclusive basis and shall be deemed to be followed by the words "without limitation";
- (nn) "Indebtedness" means the aggregate of all debts, liabilities and obligations owing under the Arnaki Loan, inclusive of all interest, costs, fees, expenses and amounts that may accrue in accordance with the Arnaki Loan and Arnaki Loan Contracts;
- (oo) "Intellectual Property" means any, and all, of the following in any jurisdiction throughout the world:
 - (i) trademarks, including all applications and registrations and the goodwill connected with the use and symbolized by the foregoing;

- (ii) copyrights and industrial designs, including all applications and registrations relating to the foregoing;
- (iii) trade secrets and confidential know-how;
- (iv) patents and patent applications;
- (v) websites and internet domain name registrations; and
- (vi) other intellectual property and related proprietary rights, interests and protections.
- (pp) "Liabilities" means liabilities, obligations or commitments of any nature whatsoever, whether asserted or unasserted, known or unknown, absolute or contingent, accrued or unaccrued, matured or unmatured or otherwise.
- "Material Adverse Change" or "Material Adverse Effect" means any change, (qq)development, effect, event, circumstance, fact or occurrence that, individually or in the aggregate with such other changes, developments, effects, events, circumstances, facts or occurrences, is, or would reasonably be expected to be, material and adverse to the properties, assets, liabilities (contingent or otherwise), condition (financial or otherwise), operations or results of operations of the Business; other than any change, development, effect, event, circumstance, fact or occurrence arising out of, attributable to or resulting from: (A) any action expressly required or permitted by this Agreement or relating to the Debtor's current financial condition, including the Receivership Proceedings; (B) general political, economic or financial conditions in Canada or elsewhere in the world; (C) any change generally affecting the industries in which the Business is conducted (including changes in general market prices, or regulatory changes in any such industry); (D) acts of terrorism or war (whether or not declared); (E) any changes to existing Applicable Law (including the interpretation thereof); (F) any changes to IFRS or the adoption, implementation or proposal of any new accounting principles; (G) hurricanes, earthquakes, storms, floods or other natural disasters, epidemics, pandemics (including COVID), outbreak or escalation of hostilities, the declaration of war, acts of terrorism, or acts of God; (H) any action consented to by the Buyer; (I) any failure by the Debtor to meet any projections or estimates (including internal projections or estimates) of revenues, earnings, working capital or performance for any period; or (J) any action, change, development, effect, event, circumstance, fact or occurrence that is attributable to or otherwise caused by the Seller, the Debtor or the Buyer;
- (rr) "Parties" means the Seller and the Buyer collectively, and "Party" means either the Seller or the Buyer; provided, however that, Parties and Party may also refer to the Debtors, as applicable;
- (ss) "Permitted Encumbrances" means those permitted encumbrances (if any) provided for in the Vesting Order;
- (tt) "Person" means any individual, partnership, limited partnership, limited liability company, joint venture, syndicate, sole proprietorship, company or corporation with or without share capital, unincorporated association, trust, trustee, executor, administrator or other legal personal representative, Governmental Authority or other entity however designated or constituted;
- (uu) "Priority Charges" means any amounts that rank in priority to the Arnaki Security, including without limitation any Court ordered charges or statutory priority claims (whether arising in the Receivership Proceeding or any other proceeding involving the Debtor,

including without limitation a bankruptcy proceeding), the Receivership Costs, the Employee Priority Payments and the CRA Priority Payments;

- (vv) "Priority Payables" means any current or future amounts owing as secured by the Priority Charges;
- (ww) "Purchase Price" has the meaning given to such term in Section 3.1;
- (xx) "Purchased Assets" means those assets, undertakings and properties listed in Schedule 1.1(xx), but for greater certainty, shall not include the Excluded Assets;
- (yy) "Receiver" has the meaning given to such term in Recital B;
- (zz) "Receiver's Borrowing Charge" has the meaning given to such term in the Receivership Order;
- (aaa) "Receiver's Certificate" means the certificate filed with the Court by the Receiver certifying that the Receiver is satisfied that: (i) all conditions to Closing set forth in Article 6 have been satisfied or waived; and (ii) the Buyer has paid, and the Seller has received, the Purchase Price;
- (bbb) "Receiver's Charge" has the meaning given to such term in the Receivership Order;
- (ccc) "Receivership Costs" means all of the taxable fees, costs and expenses associated with the Receiver's administration of the Debtor's estate pursuant to the Receivership Order, including without limitation, the Receiver's Fees, Receiver's Charge and Receiver's Borrowing Charge.
- (ddd) "Receiver's Fees" means the fees and disbursements of the Receiver and the Receiver's counsel incurred in connection with the receivership of the Debtor, secured by the Receiver's Charge, as approved by the Court from time to time;
- (eee) "Receivership Order" has the meaning given to such term in Recital B;
- (fff) "Receivership Proceedings" has the meaning given to such term in Recital B;
- (ggg) "Representative" means, in respect of a Person, each director, officer, employee, agent, legal counsel, accountant, consultant, contractor, professional advisor and other representative of such Person and its affiliates;
- (hhh) "Seller" has the meaning given to such term in the preamble to this Agreement;
- (iii) "SISP" means the sale and investment solicitation process set forth in Schedule "1.1(iii)" hereto;
- (jjj) "SISP Order" means an order of the Court substantially in the form attached as Schedule "1.1(jjj)" hereto that, among other things, approves the SISP and related matters, with any amendments thereto to be acceptable to each of the Seller and the Buyer, each acting reasonably.
- (kkk) "SolvAQUA Claims" means any and all claims of the Debtor, including but not limited to any claims related to alleged breaches under various contracts between SolvAQUA and each of Vivaventures Inc. and Clear Capital Partners as well as any corresponding insurance claims related to coverage provided by Export Development Canada in relation to these contracts;

(III) "Successful Bid" has the meaning ascribed to it in the SISP;

(mmm) "Tax" and "Taxes" means any and all:

- taxes, duties, fees, excises, premiums, assessments, imposts, levies and other charges or assessments of any kind whatsoever imposed by any Governmental Authority, including those with respect to income, goods and services, harmonized sales, transfer, land transfer, use, real or personal property, and registration fees; and
- (ii) all interest, penalties, fines, additions to tax or other additional amounts imposed by any Governmental Authority on or in respect of amounts of the type described in clause (i) above or this clause (ii);
- (nnn) "Tax Act" means the Income Tax Act (Canada), as amended;
- (000) **"Transaction**" means, collectively, the sale and purchase of the Purchased Assets pursuant to this Agreement and all other transactions contemplated by this Agreement that are to occur contemporaneously with the sale and purchase of the Purchased Assets;
- (ppp) "Transfer Taxes" has the meaning given to such term in Section 7.7(c);
- (qqq) "Vesting Order" means an order of the Court substantially in the same form as the Alberta Model Sale Approval and Vesting Order, with any amendments thereto to be acceptable to each of Seller and Buyer, each acting reasonably;
- (rrr) "Vivakor Equipment" means two completed wastewater management units with serial number J-888011-A; and
- (sss) "Vivakor Equipment Purchase Price" means an amount to be determined by the Equipment Owner and the Buyer, acting reasonably, for the purchase of the Vivakor Equipment; provided that it is determined by: (i) the Receiver, Buyer and Equipment Owner; or (ii) the Court, that the Vivakor Equipment is owned by a Person other than SolvAQUA.

1.2 Schedules

The schedules to this Agreement are an integral part of this Agreement.

Schedule	Description Purchased Assets	
Schedule 1.1(xx)		
Schedule 1.1(iii)	SISP	
Schedule 1.1(jjj)	Form of SISP Order	

1.3 Statutes

Unless specified otherwise, reference in this Agreement to a statute refers to that statute as it may be amended, or to any restated or successor legislation of comparable effect.

1.4 Headings and Table of Contents

The inclusion of headings and a table of contents in this Agreement is for convenience of reference only and shall not affect the construction or interpretation hereof.

1.5 Gender and Number

In this Agreement, unless the context otherwise requires, words importing the singular include the plural and vice versa and words importing gender include all genders.

1.6 Currency

Except where otherwise expressly provided, all amounts in this Agreement are stated and shall be paid in Canadian dollars.

1.7 Invalidity of Provisions

Each of the provisions contained in this Agreement is distinct and severable and a declaration of invalidity or unenforceability of any such provision or part thereof by a court of competent jurisdiction shall not affect the validity or enforceability of any other provision hereof.

1.8 Entire Agreement

This Agreement and the agreements and other documents required to be delivered pursuant to this Agreement, constitute the entire agreement between the Parties and set out all the covenants, promises, warranties, representations, conditions and agreements between the Parties in connection with the subject matter of this Agreement and supersede all prior agreements, understandings, negotiations and discussions, whether oral or written, pre-contractual or otherwise. There are no covenants, promises, warranties, representations, conditions, understandings or other agreements, whether oral or written, pre-contractual or otherwise. There are no covenants, promises, warranties, representations, conditions, understandings or other agreements, whether oral or written, pre-contractual or otherwise, express, implied or collateral between the Parties in connection with the subject matter of this Agreement except as otherwise specifically set forth in this Agreement and any document required to be delivered pursuant to or in respect of this Agreement.

1.9 Waiver, Amendment

Except as expressly provided in this Agreement, no amendment or waiver of this Agreement shall be binding unless executed in writing by all Parties hereto. No waiver of any provision of this Agreement shall constitute a waiver of any other provision nor shall any waiver of any provision of this Agreement constitute a continuing waiver unless otherwise expressly provided.

1.10 Governing Law; Jurisdiction and Venue

This Agreement, the rights and obligations of the Parties under this Agreement, and any claim or controversy directly or indirectly based upon or arising out of this Agreement or the Transaction (whether based on contract, tort, or any other theory), including all matters of construction, validity and performance, shall in all respects be governed by, and interpreted, construed and determined in accordance with, the laws of the Province of Alberta and the federal laws of Canada applicable therein, without regard to the conflicts of law principles thereof. The Parties consent to the exclusive jurisdiction and venue of the Court prior to a Final order of the Court terminating the Receivership Proceedings and thereafter to the Courts of Alberta for the resolution of any disputes arising under this Agreement. Each Party agrees that service of process on such Party as provided in Section 11.6 shall be deemed effective service of process on such Party.

ARTICLE 2 PURCHASE AND SALE

2.1 Agreement to Purchase and Sell Purchased Assets

- (a) Upon and subject to the terms and conditions of this Agreement at Closing and effective as of the Closing Time the Seller shall sell to the Buyer, and the Buyer shall purchase from the Seller, free and clear of all Encumbrances, the Purchased Assets.
- (b) Provided that Closing occurs, possession risk, beneficial and legal ownership of the Purchased Assets shall, transfer from the Seller to the Buyer in accordance with the Vesting Order and the terms of this Agreement.

2.2 "As is, Where Is"

The Buyer acknowledges and agrees that all of the Purchased Assets are being purchased on an "as is, where is" basis as they shall exist at Closing. Unless and solely to the extent expressly set forth in this Agreement, no representation, warranty or covenant is expressed or implied by the Seller, including any warranties as to title, encumbrances, description, merchantability or fitness for a particular purpose, environmental compliance, condition, quantity or quality, or in respect of any other matter or thing whatsoever concerning the Business and/or the Purchased Assets or the right of the Seller to sell or assign the same. Without limiting the generality of the foregoing, any and all conditions, warranties, or representations expressed or implied pursuant to the *Sale of Goods Act* (Alberta), as amended, or similar legislation do not apply hereto and have been waived by the Buyer. This Section 2.2 shall not merge on Closing and is deemed incorporated by reference into all Closing Documents and deliveries.

ARTICLE 3 PURCHASE PRICE AND RELATED MATTERS

3.1 Purchase Price

The purchase price payable by the Buyer to the Seller for the Purchased Assets (the "Purchase Price") shall be the aggregate sum of:

- (a) a cash payment of the Cash Component; and
- (b) the balance of the Purchase Price, equal to the amount of \$2,500,000, to be paid by way of set off against (as a non-cash credit reduction of) the Indebtedness (the "Credit Bid Component").

3.2 Deposit

- (a) The Deposit shall be paid by wire transfer in immediately available funds within two (2) Business Days of the SISP Order having been granted by the Court. The Deposit shall be held, pending Closing, by the Seller's counsel in a non-interest bearing account.
- (b) If the Closing does not occur solely by reason of the default of the Buyer, the full amount of the Deposit, less any applicable withholding tax, shall become the property of and be retained by the Seller to compensate it for expenses incurred in connection with the Transaction contemplated in this Agreement and the delay caused to the Seller's efforts to sell the Purchased Assets. The entitlement of the Seller to retain the Deposit in such circumstances shall not limit the Seller's right to exercise any other rights which the Seller may have against the Buyer in respect of such default.

(c) If the Closing does not occur for any reason other than the default of the Buyer, the full amount of the Deposit, less any applicable withholding tax, shall be returned forthwith to the Buyer. It is understood that non-completion of the transaction of purchase and sale contemplated by this Agreement by the Buyer because of the non-fulfillment of any conditions precedent set out in Section 6.1 (where such condition precedent was in favour of the Buyer) or 6.2 shall not be considered to be a default of the Buyer.

3.3 Satisfaction of Purchase Price

The Buyer shall satisfy the Purchase Price at the Closing Time by:

- (a) irrevocably and unconditionally authorizing the Seller's counsel to release the Deposit to the Seller, and the Buyer paying to the Receiver in trust the Cash Component (less the Deposit) by wire transfer in immediately available funds pursuant to wire transfer instructions to be provided by the Receiver to the Buyer no later than one (1) Business Day prior to the Closing Date; and
- (b) delivering an acknowledgement signed by the Buyer and Arnaki that the Indebtedness has been reduced by the amount of \$2,500,000 in consideration of the balance of the Purchase Price.

3.4 Purchase Price Allocation

The Purchase Price shall be allocated among the Purchased Assets as agreed by the Parties herein and/or in separate written instruments, acting reasonably. Such allocation shall be binding and the Buyer and the Seller shall report the purchase and sale of the Purchased Assets and file all filings which are necessary or desirable under the Tax Act and any other applicable Tax legislation to give effect to such allocations and shall not take any position or action inconsistent with such allocation.

3.5 Payment of Certain Liabilities

Upon payment of the Cash Component to the Seller, the Seller shall satisfy all known Priority Payables as required to be paid on Closing, in accordance with the Vesting Order.

3.6 Adjustments

The Parties will make, without duplication, the usual adjustments relating to the Purchased Assets, as of the Closing Date.

3.7 Determination of Final Purchase Price

In addition to those adjustments described in Section 3.6, the Parties acknowledge that adjustments will be required after the Closing Date to finally determine the Purchase Price once the final amount of the Priority Payables is known. The Purchase Price shall be finally determined on the date that the Receiver is discharged pursuant to a Discharge Order issued by the Court, or such other date as the parties agree to in writing ("Adjustment Date"). On the Adjustment Date, the Parties shall agree on the final Purchase Price and the Holdback Amount shall be distributed by the Seller as follows:

- Seller shall pay from the Holdback any Priority Payables not previously satisfied under Section 3.5;
- (b) following the payment of any Priority Payables under Section 3.5 and Section 3.7(a), any remainder of the Holdback Amount shall be reimbursed to the Buyer; and

(c) in the event the Holdback Amount is not sufficient to satisfy the Priority Payables, the Buyer shall pay to the Seller within two (2) Business Days of being notified of such deficiency, the deficiency by way of certified cheque, bank draft or guaranteed electronic funds.

ARTICLE 4 REPRESENTATIONS AND WARRANTIES BY THE SELLER

The Seller represents and warrants to the Buyer and acknowledges that the Buyer is relying upon the following representations and warranties in connection with its purchase of the Purchased Assets the matters set out below:

4.1 Validly Appointed

The Seller has been validly appointed by the Court as receiver and manager of all of the assets, undertakings and properties of the Debtor acquired for, or used in relation to a business carried on by the Debtor, including all proceeds thereof, pursuant to the Receivership Order.

4.2 Recitals

The Recitals hereto are true and correct as of the date of this Agreement.

4.3 Due Authorization and Enforceability of Obligations

Subject to Court Approval being obtained, the Seller has all necessary power, authority and capacity to enter into this Agreement and to carry out its obligations under this Agreement. This Agreement has been, and at the Closing Time the Closing Documents will be, duly executed and delivered by the Seller in its capacity as Receiver.

4.4 Residence of the Seller

The Seller is not a non-resident of Canada within the meaning of section 116 of the Tax Act.

4.5 No Other Representations, Warranties or Covenants

All of the Purchased Assets are being purchased on an "as is where is" basis. Unless and solely to the extent expressly set forth in this Agreement, no representation, warranty or covenant is expressed or implied by the Seller, including any warranties as to title, encumbrance, description, merchantability or fitness for a particular purpose, environmental compliance, condition, quantity or quality, or in respect of any other matter or thing whatsoever concerning the Business and/or the Purchased Assets or the right of the Seller to sell or assign the same. The disclaimer in this Section 4.5 is made notwithstanding the delivery or disclosure to the Buyer or its directors, officers, employees, agents or representatives of any documentation or other information.

ARTICLE 5 REPRESENTATIONS AND WARRANTIES OF THE BUYER

The Buyer represents and warrants to the Seller as follows, and acknowledges that the Seller is relying upon the following representations and warranties in connection with its sale of the Purchased Assets:

5.1 Existence

The Buyer is duly organized and validly existing under the laws of its jurisdiction of organization.

5.2 Recitals

The Recitals hereto are true and correct as of the date of this Agreement.

5.3 Due Authorization

The Buyer has all necessary power, authority and capacity to enter into this Agreement and to carry out its obligations under this Agreement. The execution and delivery of this Agreement and the consummation of the Transaction have been duly authorized by all necessary action of the Buyer. This Agreement has been duly executed and delivered by the Buyer and constitutes a valid and binding obligation of the Buyer enforceable against it in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, moratorium, reorganization and similar laws affecting creditors generally and by general principles of equity.

5.4 Absence of Conflicts

The Buyer is not a party to, bound or affected by or subject to any provision in its articles, by-laws or other constating documents or Applicable Laws or Governmental Authorizations, approvals, franchises, orders, certificates, consents, directives, notices, licences, permits, variances, registrations or other rights issued, granted or given by or from any Governmental Authority that would be violated, breached by, or under which any default would occur or with notice or the passage of time would, be created as a result of the execution and delivery of, or the performance of obligations under, this Agreement or any other agreement or document to be entered into or delivered under the terms of this Agreement, except for any violations, breaches or defaults or any Applicable Laws or any Governmental Authorizations, approvals, orders, certificates, consents, directives, notices, licences, permits, variances, registrations or other rights issued, granted or given by or from any Governmental Authority, that would not have a material and adverse effect on the ability of the Buyer to consummate the transactions hereunder.

5.5 No Litigation

Other than the Receivership Proceedings, there is no Claim that is pending or, to the Buyer's knowledge, threatened in any court or before any Governmental Authority that would adversely affect the Buyer's ability to perform its obligations under this Agreement on a timely basis.

5.6 Approvals and Consents

Except for Court Approval, no authorization, consent or approval of, or filing with or notice to, any Governmental Authority or any other Person is required in connection with the execution, delivery or performance by the Seller of any of its covenants and obligations under this Agreement and each of the Closing Documents to be executed and delivered by the Seller hereunder, including the purchase of any of the Purchased Assets hereunder.

5.7 GST Registration

The Buyer, or its nominee acquiring the Purchased Assets, is, or at the Closing Time will be, registered for purposes of the GST Legislation and will provide its registration number to the Seller.

5.8 Acquisition of Business

Buyer is acquiring under this Agreement all or substantially all the property that can reasonably be regarded as being necessary for it to carry on the Business as a business.

5.9 Financial Ability

The Buyer has and will have at all relevant times, the financial ability and sufficient funds to perform all of its obligations under this Agreement, and the availability of such funds will not be subject to the consent, approval or authorization of any Person or the availability of any financing.

5.10 No Additional Due Diligence

The Buyer acknowledges and agrees that: (a) it has had an opportunity to conduct any and all due diligence regarding the Purchased Assets and the Business prior to the execution of this Agreement; (b) it has relied solely upon its own independent review, investigation and/or inspection of any documents and/or the Purchased Assets and/or the Business; (c) it is not relying upon any written or oral statements, representations, promises, warranties or guaranties whatsoever, whether express or implied, (by operation of law or otherwise), regarding the Purchased Assets or the Business or the completeness of any information provided in connection therewith, except as expressly stated in this Agreement; and (d) the obligations of the Buyer under this Agreement are not conditional upon any additional due diligence.

ARTICLE 6 CONDITIONS

6.1 Conditions for the Benefit of the Buyer and the Seller

The respective obligations of the Buyer and of the Seller to consummate the Transaction are subject to the satisfaction of, or compliance with, at or prior to the Closing Time, each of the following conditions:

- no provision of any Applicable Law and no judgment, injunction, order or decree that prohibits the consummation of the purchase of the Purchased Assets pursuant to this Agreement shall be in effect;
- (b) the Receivership Order and the SISP Order shall have been issued and entered, and shall be Final, and the Receivership Proceedings shall be ongoing;
- the Stalking Horse Bidder shall be the Successful Bidder (each, as defined in the SISP) under the SISP;
- (d) the Vesting Order shall have been issued and entered on or before November 30, 2022 or on or before such later date as the Parties agree to in writing, and shall be Final.

6.2 Conditions for the Benefit of the Buyer

The obligation of the Buyer to consummate the Transaction is subject to the satisfaction of, or compliance with, or waiver by the Buyer of, at or prior to the Closing Time, each of the following conditions (each of which is acknowledged to be for the exclusive benefit of the Buyer):

(a) the representations and warranties of the Seller set forth in this Agreement shall be true and correct at the Closing Time with the same force and effect as if made at and as of such time, except where any failure or failures of any such representations and warranties to be so true and correct would not, individually or in the aggregate cause a Material Adverse Change (and, for this purpose, any reference to "material", "Material Adverse Change" or any other concept of materiality in such representations and warranties shall be ignored);

- (b) the covenants contained in this Agreement to be performed by the Seller at or prior to the Closing Time shall have been performed in all material respects as at the Closing Time;
- (c) the Buyer shall have received a certificate confirming the satisfaction of the conditions contained in Sections 6.2(a) and 6.2(b), signed for and on behalf of the Seller without personal liability by an authorized signatory of the Seller or other Persons reasonably acceptable to the Buyer, in each case in form and substance reasonably satisfactory to the Buyer;
- (d) the Closing Documents, all other documents relating to the due authorization and completion of the Transaction and all actions and proceedings taken on or prior to the Closing in connection with the performance by the Seller of its obligations under this Agreement shall be satisfactory to the Buyer, acting reasonably, and the Buyer shall have received copies of all such documents and evidence that all such actions and proceedings have been taken as it may reasonably request in form and substance reasonably satisfactory to the Buyer;
- (e) the Purchased Assets shall be assigned and transferred to the Buyer free and clear of all Encumbrances, in accordance with the Vesting Order;
- (f) since the date first written above, no Material Adverse Change shall have occurred.

6.3 Conditions for the Benefit of the Seller

The obligation of the Seller to consummate the Transaction is subject to the satisfaction of, or compliance with, or waiver where applicable, by the Seller of, at or prior to the Closing Time, each of the following conditions (each of which is acknowledged to be for the exclusive benefit of the Seller):

- (a) the representations and warranties of the Buyer set forth in this Agreement shall be true and correct in all material respects at the Closing Time with the same force and effect as if made at and as of such time, except where any failure or failures of any such representations and warranties to be so true and correct would not, individually or in the aggregate cause a Material Adverse Change (and, for this purpose, any reference to "material", "Material Adverse Change" or any other concept of materiality in such representations and warranties shall be ignored);
- (b) the covenants contained in this Agreement to be performed by the Buyer at or prior to the Closing Time, including payment of the Purchase Price, shall have been performed in all material respects as at the Closing Time;
- (c) the Seller shall have received a certificate confirming the satisfaction of the conditions contained in Sections 6.3(a) and 6.3(b) signed for and on behalf of the Buyer without personal liability by an executive officer of the Buyer or other persons reasonably acceptable to the Seller, in each case in form and substance reasonably satisfactory to the Seller;
- (d) all other Closing Documents required pursuant to this Agreement to be delivered by the Buyer on Closing in form and substance reasonably satisfactory to the Seller;
- (e) the Buyer shall have delivered to the Seller the Deposit as required under Section 3.2;

- (f) the Buyer shall have delivered to the Seller all of the deliverables contained in Section 10.2 in form and substance satisfactory to the Seller acting in its sole discretion;
- (g) the Buyer shall have obtained an acknowledgment from Arnaki confirming that the Indebtedness has been reduced by the amount of \$2,500,000 in consideration of the balance of the Purchase Price; and
- (h) the Buyer shall have delivered and released to the Seller the acknowledgment required under Section 3.3(b) confirming that the Indebtedness has been reduced by the amount of \$2,500,000 in consideration of the balance of the Purchase Price.

ARTICLE 7 ADDITIONAL AGREEMENTS OF THE PARTIES

7.1 Access to Information

Until the Closing Time, the Seller and/or the Debtor shall furnish the Buyer's personnel engaged in the Transaction and their accountants, legal advisers, consultants and representatives during normal business hours, all such information relating to the Purchased Assets as the Buyer may reasonably request in connection with the Transaction. Notwithstanding the foregoing, the Seller and/or the Debtor shall not be required to disclose any information, records, files or other data to the Buyer where prohibited by any Applicable Laws or such disclosure would have the effect of causing the waiver of any solicitor-client privilege.

7.2 Conduct of Business Until Closing Time

Except: (1) as expressly required by this Agreement; and (2) with the prior written consent of the Buyer (not to be unreasonably withheld, conditioned or delayed), as necessary or advisable in connection with or pursuant to the Receivership Proceedings, or as required by Applicable Law (including to the extent required by any Contract), to the extent reasonably necessary, and as practicable having regard to the Receivership Proceedings and any Court order issued in the Receivership Proceedings, the Seller shall:

- (a) not amend, terminate or assign any Contracts that are included in the Purchased Assets and material to the Business;
- (b) not waive or assign any material rights of the Business under any Contract included in the Purchased Assets and material to the Business; and
- (c) not enter into any lease, contract or agreement, licence or other commitment related to the Business that would constitute a personal property lease, real property lease or Contract except, in each case, in the ordinary course of the Business.

7.3 Further Assurances

Each of the Parties hereto shall promptly do, make, execute or deliver, or cause to be done, made, executed or delivered, all such further acts, documents and things as the other Parties hereto may reasonably require from time to time for the purpose of giving effect to this Agreement and shall use commercially reasonable efforts and take all such steps as may be reasonably within its power to implement to their full extent the provisions of this Agreement. Upon and subject to the terms and conditions of this Agreement and subject to the directions of any applicable courts to the Seller, the Parties shall use their commercially reasonable efforts to take or cause to be taken all actions and to do or cause to be done all things necessary proper or advisable under Applicable Laws to consummate and make effective the Transaction, including using commercially reasonable efforts to satisfy the conditions precedent to the obligations of the Parties hereto.

7.4 Tax Matters

- (a) The Buyer and the Seller agree to use commercially reasonable efforts to furnish or cause to be furnished to each other, as promptly as practicable, such information and assistance relating to the Purchased Assets as is reasonably necessary for the preparation and filing of any Tax return, claim for refund or other required or optional filings relating to Tax matters, for the preparation for and proof of facts during any Tax audit, for the preparation for any Tax protest, for the prosecution of any suit or other proceedings relating to Tax matters and for the answer to any governmental or regulatory inquiry relating to Tax matters.
- (b) For purposes of any income Tax return related to the Transaction, the Buyer and, to the extent applicable, the Seller, agree to report the Transaction in a manner consistent with the Purchase Price allocation determined in accordance with Section 3.4, and the Buyer and the Seller shall not voluntarily take any action inconsistent therewith in any such Tax return, refund claim, litigation or otherwise, unless required by applicable Tax laws. The Buyer and the Seller shall each be responsible for the preparation of their own statements required to be filed under the Tax Act and other similar forms in accordance with applicable Tax laws.
- (c) All amounts payable by the Buyer to the Seller pursuant to this Agreement are exclusive of any GST, or any other federal, provincial, state or local or foreign value-added, sales, use, consumption, multi-staged, ad valorem, personal property, customs, excise, stamp, transfer, land or real property transfer, or similar Taxes, duties, or charges, or any recording or filing fees or similar charges (collectively, "Transfer Taxes"). All Transfer Taxes are the responsibility of and for the account of the Buyer. The Buyer and the Seller agree to cooperate to determine the amount of Transfer Taxes payable in connection with the Transaction. If the Seller is required by Applicable Law or by administration thereof to collect any applicable Transfer Taxes from the Buyer, the Buyer shall pay such amounts to the Seller concurrent with the payment of any consideration payable pursuant to this Agreement, and the Seller shall remit or account for such Transfer Taxes to the applicable Governmental Authority on a timely basis and otherwise in accordance with Applicable Laws.
- (d) The Seller and the Buyer shall jointly make, and the Buyer will file, the election provided for under GST Legislation such that no GST will be payable with respect to the purchase and sale of any of the Purchased Assets. The Buyer shall at all times indemnify and hold harmless the Seller and their directors, officers and employees, against and in respect of any and all amounts assessed by the Minister of National Revenue (Canada) as a consequence of such Minister determining, for any reason, that the election is unavailable, inapplicable, invalid or not properly made.

7.5 Fees and Expenses

Except as expressly provided in this Agreement, all fees and expenses incurred in connection with the negotiation and settlement of this Agreement and the completion of the Transaction, including the fees and disbursements of counsel, financial advisors and accountants, shall be paid by the Person incurring such fees or expenses.

7.6 Advice and Direction

The Parties acknowledge that the Seller is entitled (but not required) to seek the advice and directions of the Court in respect of any determination to be made, consent right to be exercised or other action to be taken by the Seller under this Agreement.

ARTICLE 8 COURT ORDER

8.1 Vesting Order

- (a) The Seller shall file an application with the Court for the issuance of the Vesting Order, which Vesting Order shall only be effective upon the Buyer being determined to be the Successful Bidder and the Receiver filing a certificate substantially in the form attached to the SISP Order. Such application shall be scheduled for a date that is on or before December 14, 2022, or such later date as may be scheduled upon the agreement of the Parties (acting reasonably), due to court availability or pursuant to direction of the Court.
- (b) The Buyer shall cooperate with the Seller acting reasonably, as may be necessary, in obtaining the Vesting Order.
- (c) The Seller shall use its best efforts to obtain the Vesting Order as promptly as practicable.

ARTICLE 9 TERMINATION

9.1 Termination

This Agreement may be terminated at any time prior to Closing as follows:

- (a) subject to any approvals required from the Court or otherwise pursuant to the Receivership Proceedings, by mutual written consent of the Seller and the Buyer;
- (b) by either Party, upon written notice to the other, if a Governmental Authority issues an order prohibiting the transactions contemplated hereby, which order shall have become final and non-appealable;
- by either Party, if this Agreement is not the Successful Bid (as determined pursuant to the SISP);
- (d) by either Party upon denial of the SISP Order or the Vesting Order (or if any such order is stayed, vacated or varied with the consent of the Parties);
- (e) by the Seller upon written notice to the Buyer if there has been a material violation or breach by the Buyer of any covenant, representation or warranty which would prevent the satisfaction of any condition set forth in Section 6.1 or 6.3 on the Closing Date and such violation or breach has not been waived by the Seller or cured within five (5) days after written notice thereof from the Seller, unless the Seller is in material breach of their obligations under this Agreement; and
- (f) by the Buyer upon written notice to the Seller, if there has been a material violation or breach by the Seller of any covenant (save and except for the covenants set out in Section 7.2), representation or warranty which would prevent the satisfaction of any condition set forth in Section 6.1 or 6.2 on the Closing Date and such violation or breach has not been waived by the Buyer or cured within five (5) days after written notice thereof from the Buyer, unless the Buyer is in material breach of their obligations under this Agreement.

9.2 Effect of Termination

In the event of termination of this Agreement pursuant to Section 9.1, this Agreement shall forthwith become null and void, except as set forth in Sections 3.2 and Article 11, and nothing herein shall relieve

any Party from liability for any breach of this Agreement, or to impair the right of any Party to compel specific performance by any other Party of its obligations under this Agreement.

9.3 Expense Reimbursement and Transaction Fee

- (a) If the Buyer is not the Successful Bidder (as defined in the SISP) pursuant to the SISP, the Buyer shall be entitled to reimbursement for its expenses and a transaction fee in connection with the Transaction in the aggregate amount of \$175,000 the "Expense Reimbursement and Transaction Fee"). Such amount shall be payable by the Successful Bidder to the Buyer in immediately available funds to an account designated by the Buyer concurrently with such closing.
- The payment of the Expense Reimbursement and Transaction Fee shall be approved in (b) the Vesting Order, as contemplated by the SISP, and shall be made in priority to amounts secured by existing security, except as provided in the Vesting Order contemplated by the SISP. Each of the Parties acknowledges that the agreements contained in this Section 9.3 are an integral part of the Transaction and that, without those agreements, the Parties would not enter into this Agreement. The Parties further acknowledge and agree that the Expense Reimbursement and Transaction Fee is a payment of liquidated monetary damages which are a genuine pre-estimate of the costs and damages which the Buyer will suffer or incur as a result of the non-completion of this Agreement, that such payment is not for repayment of current Indebtedness, lost profits or a penalty, and that no Party shall take any position inconsistent with the foregoing. The Seller irrevocably waives any right it may have to raise as a defense that any such liquidated damages are excessive or punitive. Each of the Parties hereby acknowledge and agree that, under circumstances where the Buyer is entitled to the Expense Reimbursement and Transaction Fee and such amounts are paid in full to the Buyer, the Buyer shall be precluded from any other remedy against the Seller at law or in equity or otherwise in respect of damages for non-completion of this Agreement, and in any such case it shall not seek to obtain any recovery, judgment, or damages of any kind, including consequential, indirect, or punitive damages, against the Seller or any of its respective directors, officers, employees, partners, managers, members, shareholders or affiliates in connection with the non-completion of this Agreement or the Transaction.
- (c) Subject to the last sentence of the preceding paragraph, nothing in this Section 9.3 shall preclude any Party from seeking injunctive relief to restrain any breach or threatened breach of the covenants or agreements set forth in this Agreement or otherwise to obtain specific performance of any such covenants or agreement, and any requirement for securing or posting of any bond in connection with the obtaining of any such injunction or specific performance is hereby being waived.

ARTICLE 10 CLOSING

10.1 Location and Time of Closing

The Closing shall take place at the Closing Time on the Closing Date at the Calgary, Alberta offices of Cassels Brock & Blackwell LLP, or at such other location as may be agreed upon by the Parties hereto, which may include by virtual means.

10.2 Closing Deliveries

(a) At the Closing, the Seller shall deliver to the Buyer the documents required to be delivered by the Seller pursuant to Sections 6.2 and Article 7.

- (b) At the Closing, the Buyer shall deliver to the Seller:
 - (i) the Cash Component, less the Deposit amount;
 - (ii) a duly executed election pursuant to GST Legislation and any certificates, elections or other documents required to be delivered pursuant to Article 7.
 - (iii) the documents required to be delivered by the Buyer pursuant to Section 6.3; and
 - (iv) any other documents reasonably requested by the Seller in order to effect or evidence the consummation of the Transaction or otherwise provided for under this Agreement.

10.3 Receiver's Capacity

The Parties hereby acknowledge and agree that the Receiver is entering into this agreement in its capacity as Court-appointed receiver of the Debtor, and not in its personal capacity.

ARTICLE 11 GENERAL MATTERS

11.1 Confidentiality

- (a) Except to the extent otherwise specifically provided in this Section 11.1, each Party, on behalf of itself and its affiliates, agrees to keep the other Party's Confidential Information confidential and not to use the other Party's Confidential Information in any manner except as required to perform the obligations set out in this Agreement. Each Party agrees to be responsible for any breach of this Section 11.1 by any of its affiliates and its and their respective directors, employees, advisors, agents and representatives.
- (b) Notwithstanding anything to the contrary herein, each Party maintains the right to disclose the other Party's Confidential Information if required to do so by Applicable Laws or requirement of a Governmental Authority, or to appropriate Tax authorities in order to describe the tax treatment and tax structure of the Transaction; provided that the disclosure of such Confidential Information will be limited only to that purpose and provided further that it will use reasonable efforts to cooperate with the other Party in limiting the disclosure of the Confidential Information.
- (c) At the other Party's request, a Party will destroy all of the other Party's Confidential Information, provided that it is permitted to retain one copy of any Confidential Information to the extent required by Applicable Laws or its internal record keeping policies.
- (d) Any Confidential Information of the Seller that constitutes part of the Purchased Assets will cease to be Confidential Information of the Seller and will become Confidential Information of the Buyer on Closing.

11.2 Public Notices

No press release or other announcement concerning the Transaction shall be made by the Seller or by the Buyer without the prior consent of the other (such consent not to be unreasonably withheld); provided, however, that subject to the last sentence of this Section 11.2, any Party may, without such consent, make such disclosure if the same is required by Applicable Law (including the Receivership Proceedings) or by any insolvency or other court or securities commission or other similar regulatory authority having jurisdiction over such Party or any of its affiliates, and, if such disclosure is required, the Party making such disclosure shall use commercially reasonable efforts to give prior oral or written notice to the other, and if such prior notice is not possible, to give such notice immediately following the making of such disclosure. Notwithstanding the foregoing: (i) this Agreement may be filed by the Seller with the Court; and (ii) the Transaction may be disclosed by the Seller to the Court, subject to redacting confidential or sensitive information as permitted by Applicable Laws. The Parties further agree that:

- (a) the Seller may prepare and file reports and other documents with the Court containing references to the Transaction and the terms thereof; and
- (b) the Seller and their professional advisors may prepare and file such reports and other documents in the Receivership Proceedings containing references to the Transaction and the terms thereof as may reasonably be necessary to complete the Transaction or to comply with their obligations in connection therewith.

Each of the Parties may issue a press release announcing the execution and delivery of this Agreement, in form and substance mutually agreed to by all of the Parties.

11.3 Survival

The representations and warranties of the Seller in this Agreement or in any agreement, document or certificate delivered pursuant to or in connection with this Agreement or the Transaction are set forth solely for the purpose of Section 6.2 and none of them shall survive Closing. The Seller shall have no liability, whether before or after the Closing, for any breach of the Seller's representations, and the Buyer acknowledges that its exclusive remedy for any such breach shall be termination of this Agreement prior to the Closing (but only if permitted by Section 9.1).

11.4 Non-Recourse

No past, present or future director, officer, employee, incorporator, member, partner, stockholder, affiliate, agent, attorney or representative of the respective Parties hereto, in such capacity, shall have any liability for any obligations or liabilities of the Buyer, the Debtors or the Seller, as applicable, under this Agreement or for any claim based on, in respect of, or by reason of, the Transaction.

11.5 Assignment; Binding Effect

No Party may assign its right or benefits under this Agreement without the consent of the other Party hereto, except that without such consent the Buyer may: (i) assign any or all of its rights and obligations hereunder to one or more of its subsidiaries or affiliates; or (ii) direct that title to all or some of the Purchased Assets be transferred to one or more of its subsidiaries or affiliates, provided that no such assignment or direction shall relieve the Buyer of its obligations hereunder; provided further that if the Buyer shall have assigned all of its rights and obligations hereunder the Buyer shall, immediately following the Closing, be deemed fully released from all of the Buyer's obligations hereunder. References to the Buyer's status under the *Investment Canada Act* are references to the ultimate Buyer. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective permitted successors and permitted assigns. Nothing in this Agreement shall create or be deemed to create any third Person beneficiary rights in any Person or entity not a Party to this Agreement other than the express third party beneficiaries of Section 11.4 hereof.

11.6 Notices

Any notice, request, demand or other communication required or permitted to be given to a Party pursuant to the provisions of this Agreement will be in writing and will be effective and deemed given under this Agreement on the earliest of: (i) the date of personal delivery; (ii) the date of transmission by email, with confirmed transmission and receipt (if sent during normal business hours of the recipient, if not, then on the next Business Day); (iii) two days after deposit with a nationally-recognized courier or overnight service such as Federal Express; or (iv) five days after mailing via certified mail, return receipt requested.

All notices not delivered personally or by facsimile will be sent with postage and other charges prepaid and properly addressed to the Party to be notified at the address set forth for such Party:

(a) If to the Buyer at:

2464525 Alberta Ltd. c/o Tingle Merrett LLP 1250, 639 - 5th Avenue S.W. Calgary, AB T2P 0M9

Attention: Scott Reeves Email: sreeves@tinglemerrett.com

(b) If to the Seller at:

MNP Ltd. 1500, 640 – 5th Avenue SW Calgary, AB T2P 3G4

Attention: Vanessa Allen Email: vanessa.allen@mnp.ca

with a copy to:

Cassels Brock & Blackwell LLP Bankers Hall West 3810, 888 3rd Street SW Calgary, AB T2P 5C5

Attention: Jeffrey Oliver Email: joliver@cassels.com

Any Party may change its address for service from time to time by notice given in accordance with the foregoing and any subsequent notice shall be sent to such Party at its changed address.

11.7 Counterparts; Facsimile Signatures

This Agreement may be signed in counterparts and each of such counterparts shall constitute an original document and such counterparts, taken together, shall constitute one and the same instrument. Execution of this Agreement by any of the Parties hereto may be evidenced by facsimile, scanned e-mail or internet transmission copy of this Agreement bearing such signature which, for all purposes, shall be deemed to be an original signature.

[Remainder of page intentionally blank]

IN WITNESS WHEREOF the Parties hereto have executed this Agreement as of the date first written above.

MNP LTD., solely in its capacity as receiver of the assets, undertakings and properties of **SOLVAQUA INC.** and not in its personal capacity

By:

Name: Vanessa Allen Title: Senior Vice President

2464525 ALBERTA LTD.

By:



Name: Scott Reeves Title: Director

Schedule 1.1(xx) Purchased Assets

All of the Debtor's right, title and interest in, to and under all of the tangible and intangible assets, properties and rights of every kind and nature and wherever located, other than the Excluded Assets, which relate to or are used or held for use in connection with the Business, including the following:

- (a) the Vivakor Equipment, unless the Buyer elects not to purchase the Vivakor Equipment in the event it is determined by (i) the Receiver, Buyer and Equipment Owner, or (ii) the Court, that the Vivakor Equipment is owned by a Person other than SolvAQUA;
- (b) the SolvAQUA Claims;
- (c) all accounts receivable, notes receivable and negotiable instruments of the Business;
- (d) all prepaid charges and expenses of the Business;
- (e) all inventory, finished goods, raw materials, work in progress, packaging, supplies, parts and other inventories of the Business;
- (f) all equipment and other tangibles assets of the Debtor, including all water treatment equipment, vehicles, tools, parts and supplies, fuel, machinery, furniture, furnishing, appliances, fixtures, office equipment and supplies, owned and licensed computer hardware and related documentation, stored data, communication equipment, trade fixtures and leasehold improvement, in each case, with any transferable warranty and service rights of the Debtor related thereto;
- (g) all Contracts;
- (h) all Intellectual Property;
- (i) all goodwill and intangibles;
- (j) all books and records;
- (k) all rights under insurance contracts and policies;
- (I) all telephone numbers, fax numbers and email addresses;
- (m) all non-disclosure agreements entered into by the Receiver on behalf of the Debtor in connection with the SISP;
- (n) all other additional assets, properties, privileges, rights and interest of the Debtor relating to the Business or the assets of the Debtor of every kind and description and wherever located, whether known or unknown, fixed or unfixed, accrued, absolute, contingent or otherwise, and whether or not specifically referred to in this Agreement; and
- (o) the Assumed Liabilities.

DocuSign Envelope ID: 8EF04403-C733-41BB-9C79-7125459F8C51

Schedule 1.1(iii) SISP

SALE AND INVESTOR SOLICITATION

INTRODUCTION

By order of the Court of Queen's Bench of Alberta (the "Court") dated August 19, 2022 (the "Appointment Order"), MNP Ltd. ("MNP") was appointed as receiver and manager (in such capacities, the "Receiver"), without security, of all of the assets, undertakings, and properties (the "Property") of SolvAQUA Inc. ("SolvAQUA" or the "Company").

THE OPPORTUNITY

It is anticipated that, on or about October 20, 2022, the Receiver will seek a Court order (the "SISP Approval Order") that will, among other things, (i) approve the terms of the binding purchase agreement (the "Stalking Horse Agreement") between the Receiver and 2464525 Alberta Ltd. (the "Stalking Horse Bidder"), and (ii) approve the stalking horse sale and investment solicitation process ("SISP") described herein.

The SISP sets out the manner in which (i) binding bids for transaction alternatives that are superior to the sale transaction to be provided for in the Stalking Horse Agreement will be solicited from interested parties, (ii) any such bid received will be addressed, (iii) any Successful Bidder will be selected, and (iv) Court approval of any Successful Bidder will be sought. Such transaction alternatives may include, among other things, the recapitalization of, investment in, arrangement of or reorganization of the Company or the business of SolvAQUA (the "Business") as a going concern or a sale of some or all of the Property or some combination thereof.

The SISP Approval Order, the procedures in respect of the SISP as contained herein (the "SISP Procedures"), a Court Order approving the transaction contemplated in the Successful Bid and vesting of the assets as contemplated therein (the "Vesting Order") and any subsequent orders issued by the Court pertaining to the SISP Procedures shall exclusively govern the process for soliciting and selecting bids and ultimately determining the Successful Bid and Successful Bidder.

The purpose of these SISP Procedures is to determine whether a transaction that is superior to the Stalking Horse Transaction can be obtained by the Receiver. For the purposes of these SISP Procedures, a "Superior Offer" shall mean:

- a credible, reasonably certain and financially viable offer made by a Qualified Bidder (as defined herein), the terms of which offer are more favourable and no more burdensome or conditional than the terms contained in the Stalking Horse Agreement; and
- an offer that provides for cash consideration in an amount equal to or greater than the Purchase Price (as defined in the Stalking Horse Agreement), plus (i) the Expense Reimbursement and Transaction Fee, (ii)

the Priority Payables, and (iii) the Holdback Amount (each as defined in the Stalking Horse Agreement). respectively.

Except to the extent otherwise set forth in a definitive agreement with a Successful Bidder, any sale of the Property or investment in SolvAQUA will be on an "as is, where is" basis and without surviving representations or warranties of any kind, nature, or description by the Receiver, or any of their respective agents, advisors or representatives, and, in the event of a sale, all of the right, title and interest of the Company in and to the Property to be acquired will be sold free and clear of all pledges, liens, security interests, encumbrances, claims, charges, options, and interests therein and thereon pursuant to a Court order, except as otherwise provided in such Court order.

TIMELINE

The following table sets out the key milestones under the SISP:

Milestones	Estimated deadlines
SISP Approval Order granted/ commencement of SISP	October 20, 2022
Distribution of SISP Notice and Teaser	October 21, 2022
Publication of Notice	October 26, 2022
Bid Deadline	November 23, 2022
Auction (if required)	By November 30, 2022
Sale Approval Hearing	By December 14, 2022
Closing Date	By December 21, 2022

The Receiver will use reasonable efforts to complete the SISP Procedures in accordance with the timelines as set out herein. The Receiver shall be permitted to make such adjustments to the timeline that it determines are reasonably necessary, acting in its sole discretion.

SOLICITATION OF INTEREST

As soon as reasonably practicable:

- the Receiver will prepare a list of potential bidders, including (i) parties that have approached the Receiver or SolvAQUA indicating an interest in the opportunity, and (ii) any local and international strategic and financial parties who the Receiver believes may be interested in purchasing all or part of the Business and Property or investing in the Company pursuant to the SISP (collectively, "Known Potential Bidders");
- 2. the Receiver will arrange for a notice of the SISP (and such other relevant information which the Receiver considers appropriate) (the "Notice") to be published in The Globe and Mail (National Edition) and any other newspaper or journal as the Receiver considers appropriate, if any; and
- the Receiver will prepare: (i) a process summary (the "Teaser") describing the opportunity, outlining the process under the SISP and inviting recipients of the Teaser to express their interest pursuant to the SISP; and (ii) a non-disclosure agreement in form and substance satisfactory to the Receiver (an "NDA").

The Receiver will publish the Notice and send the Teaser and NDA to all Known Potential Bidders beginning on October 21, 2022 and to any other party who requests a copy of the Teaser and NDA or who is identified to the Receiver as a potential bidder as soon as reasonably practicable after such request or identification, as applicable.

INTERESTED PARTIES

Access to Confidential Data Room

Any party who wishes to participate in the SISP must provide to the Receiver, unless the Receiver advises otherwise, a NDA executed by the potential bidder and a letter setting forth the identity of the potential bidder, the contact information for such potential bidder and full disclosure of the direct and indirect principals of the potential bidder.

Upon receiving a NDA from a bidder, such bidder shall be deemed to be a "Qualified Bidder". For greater certainty, no bidder shall be deemed to be a Qualified Bidder without the approval of the Receiver.

At any time prior to the Bid Deadline, the Receiver may, in its reasonable business judgment eliminate a Qualified Bidder from the SISP, in which case such Qualified Bidder will no longer be a Qualified Bidder for the purposes of the SISP.

The Receiver will provide each Qualified Bidder with access to a confidential data room providing additional information considered relevant to the Potential Transaction (the "Confidential Data Room").

The Receiver and its respective advisors make no representation or warranty as to the information contained in the Confidential Data Room or otherwise made available pursuant to the SISP, except to the extent expressly

contemplated in any definitive agreement with a Successful Bidder ultimately executed and delivered by the Receiver.

Qualified Bidders must rely solely on their own independent review, investigation and/or inspection of all information and of the Property and Business in connection with their participation in the SISP and any Transaction they enter into with the Receiver.

Due Diligence

The Receiver shall, in its reasonable business judgment and subject to competitive and other business considerations, afford each Qualified Bidder such access to due diligence materials and information relating to the Property and Business as they deem appropriate. Due diligence access may include management presentations, access to electronic data, on-site inspections, and other matters which a Qualified Bidder may reasonably request and as to which the Receiver in its reasonable business judgment may agree.

The Receiver will designate a representative to coordinate all reasonable requests for additional information and due diligence access from a Qualified Bidder and the manner in which such requests must be communicated. The Receiver will not be obligated to furnish any information relating to the Property or Business to any person. Further, and for the avoidance of doubt, selected due diligence materials may be withheld from certain Qualified Bidders if the Receiver determines such information to represent proprietary or sensitive competitive information.

FORMAL BINDING OFFERS

Qualified Bidders that wish to make a formal offer to purchase or make an investment in SolvAQUA, the Property and/or Business shall submit a binding offer (the "**Bid**") that complies with all of the following requirements to the Receiver at the addresses specified in Schedule "A" hereto (including by email or fax transmission). The Bids must be received by the Receiver by no later than 5:00 PM (Mountain Time) on November 23, 2022 (the "**Bid Deadline**"). A Bid may be an offer to acquire all, substantially all or a portion of the Property (a "**Sale Proposal**"), or to make an investment in, restructure, reorganize or refinance the Business (an "**Investment Proposal**"), or such other structure as the Bidder may propose. In order to constitute a "**Qualified Bid**", a Bid must comply with the following:

- it provides for the payment in full in cash on closing of any Priority Payables and the Holdback Amount (each as defined in the Stalking Horse Agreement), unless otherwise agreed to by the applicable holders of such Priority Charge in their sole discretion;
- it provides for the payment in full in cash on closing of the Vivakor Equipment Purchase Price (as defined in the Stalking Horse Agreement), provided that the Vivakor Equipment forms part of the purchased assets under the Bid and the Receiver advises that the Vivakor Equipment is not owned by SolvAQUA;
- 3. it provides a detailed 'sources and uses schedule' that identified, with specificity, the amount of cash consideration (the "Cash Consideration Value") and any assumptions that could reduce the net consideration payable. At a minimum, the Cash Consideration Value must be sufficient for payment in full

of the items contemplated in Sections 1 and 2 above and the Expense Reimbursement and Transaction Fee (as defined in the Stalking Horse Agreement);

- 4. in the case of a Sale Proposal, it identifies or contains the following:
 - a) the purchase price in Canadian dollars, including any requested allocation and details of any liabilities to be assumed by the Qualified Bidder;
 - b) a description of the Property that is expected to be subject to the transaction and any of the Property expected to be excluded; and
 - a specific indication of the financial capability of the Qualified Bidder and the expected structure and financing of the transaction;
- 5. in the case of an Investment Proposal, it identifies the following:
 - a) a description of how the Qualified Bidder proposes to structure the proposed investment;
 - b) the aggregate amount of the equity and/or debt investment to be made in the Business in Canadian dollars;
 - c) the underlying assumptions regarding the pro forma capital structure;
 - a specific indication of the sources of capital for the Qualified Bidder and the expected structure and financing of the transaction;
 - e) all conditions to closing that the Bidder may wish to impose; and
 - f) any other terms or conditions of the Investment Proposal that the Bidder believes are material to the transaction;
- it includes a letter stating that the Qualified Bidder's offer is binding and irrevocable until the selection of the Successful Bid (as defined below), provided that if such Bid is selected as the Successful Bid, the Bid shall remain irrevocable until the closing of the Successful Bid;
- it includes duly authorized and executed binding transaction document(s), which shall include the purchase price, investment amount and any other key economic terms expressed in Canadian dollars (the "Purchase Price"), together with all exhibits and schedules thereto;
- 8. it includes a redline to the form of transaction document(s) provided by the Receiver, if applicable;
- 9. it includes written evidence of a Qualified Bidder's ability to fully fund and consummate the proposed transaction and satisfy its obligations under the transaction documents, including a binding, irrevocable commitment for financing or other evidence of ability to consummate the proposed transaction, that will allow the Receiver to make a determination as to the Qualified Bidder's financial and other capabilities to consummate the proposed transaction;
- it does not include any request for or entitlement to any transaction fee, expense reimbursement or similar type of payment;
- 11. it is not conditional on:
 - a) approval from the Qualified Bidder's board of directors (or comparable governing body) or equity holder(s);

- b) the outcome of any further due diligence by the Qualified Bidder, apart from, if applicable, the disclosure of due diligence materials that represent proprietary or sensitive competitive information which was withheld during the due diligence process prior to the Bid Deadline; or
- c) the Qualified Bidder obtaining financing;
- 12. it fully discloses the legal name and identity and contact information of each entity or person that will be entering into the proposed transaction, that is participating in, or benefiting from, such bid, including any equity holder(s);
- 13. it includes acknowledgments and representations from the Qualified Bidder that the Qualified Bidder:
 - a) has had an opportunity to conduct any and all due diligence regarding the Property, and Business prior to making its offer (apart from, to the extent applicable, the disclosure of any due diligence materials representing proprietary or sensitive competitive information withheld during the due diligence process prior to the Bid Deadline);
 - b) has relied solely upon its own independent review, investigation and/or inspection of any documents and/or the Property in making its Bid; and
 - c) did not rely upon any written or oral statements, representations, warranties, or guarantees whatsoever, whether express, implied, statutory or otherwise, regarding the Business or the Property or the completeness of any information provided in connection therewith, except as expressly stated in the definitive transaction agreement(s);
- 14. it is accompanied by a non-refundable deposit paid by wire transfer in immediately available funds, which deposit shall be non-refundable upon the Qualified Bidder being selected as the Successful Bidder. The deposit amount shall not be less than:
 - a. 10% of the Purchase Price offered for a Sales Proposal; and
 - b. 10% of the total new investment contemplated in the Bid for an Investment Proposal.
- 15. it includes a statement that the Qualified Bidder will bear its own costs and expenses (including legal and advisor fees) in connection with the proposed transaction, and by submitting its Bid is agreeing to refrain from and waive any assertion or request for reimbursement on any basis;
- 16. it is received by the Bid Deadline;
- 17. it constitutes, in the reasonable business judgment of the Receiver, a Superior Offer; and
- it contemplates closing the Transaction set out therein on or before December 14, 2022 (the "Closing Date").

The Receiver may (in its sole discretion) waive strict compliance with any one or more of the foregoing requirements and deem such non-compliant bids to be a Qualified Bid. For greater certainty, notwithstanding the requirements specified in Section 1 to 18 above, the Stalking Horse Agreement shall be deemed to be a Qualified Bid.

SELECTION OF SUCCESSFUL BIDDERS

Following the Bid Deadline, the Receiver will assess the Bids received. The Receiver shall approve the disqualification of any Bids that are deemed not to be Qualified Bids. Only Qualified Bidders whose Bids have been designated as Qualified Bids are eligible to become the Successful Bidder(s).

The Receiver shall notify each Qualified Bidder in writing as to whether its Bid constituted a Qualified Bid within two (2) business days following the Bid Deadline, or at such later time as the Receiver deems appropriate.

If no Qualified Bids, other than the Stalking Horse Agreement have been received by the Bid Deadline, then the Receiver may consider an extension of the Bid Deadline for up to 7 business days to allow any party that submitted a Bid to consult with the Receiver and to revise their Bid such that it would provide for, or that might reasonably be expected to provide for, a Superior Offer.

Evaluation of Competing Bids

A Qualified Bid will be evaluated upon many factors, including, without limitation, items such as:

- 1. the Purchase Price and the net value provided by such Qualified Bid;
- 2. the identity, circumstances and ability of the Qualified Bidder to successfully complete their proposed transaction;
- 3. the proposed transaction documents;
- 4. factors affecting the speed, certainty and value of the proposed transaction;
- 5. the assets included or excluded from the Qualified Bid;
- 6. the likelihood and timing of consummating the proposed transaction; and
- 7. whether the proposed transaction results in a Superior Offer;

each as determined by the Receiver in its sole discretion.

Selection of Successful Bid

The Receiver shall:

- first, review and evaluate each Qualified Bid, provided that the Receiver may contact any Qualified Bidder to clarify the terms of any Bid, and the applicable Qualified Bidder may amend, modify or vary such Bid for the purpose of clarification;
- 2. second, identify if any Qualified Bid is a Superior Offer; and
- 3. third, if one or more Qualified Bids (other than the Stalking Horse Agreement) are considered to be Superior Offers, the receiver shall proceed with an auction process to determine the successful bid(s) (the

"Auction"), which Auction shall be held on or before November 30, 2022 and shall be administered in accordance with the Auction Procedures set out in Schedule "B". The successful bid(s) selected in the Auction shall constitute the "Successful Bid", and the Bidder making such Successful Bid, the "Successful Bidder".

If no Superior Offer is received by the Bid Deadline, the Auction will not be held, the Stalking Horse Bidder will be declared to be the Successful Bidder and the Receiver shall proceed to apply for the Vesting Order. Upon the granting of the Vesting Order, all Qualified Bids other than the Stalking Horse Agreement, if any, shall be deemed rejected by the Receiver on and as of the date of the filing of the Vesting Order.

If the Successful Bidder is a Qualified Bidder other than the Stalking Horse Bidder, the Receiver shall seek to finalize any remaining necessary definitive agreement(s) with respect to the Successful Bid in accordance with the relevant milestones described under the "Timeline" section of this SISP and shall apply to the Court for an order or orders (among other things) approving such Successful Bid and implementing the transaction(s) contained in such Successful Bid. All Qualified Bids other than the Successful Bid, shall be deemed rejected by the Receiver on and as of the date of approval of the Successful Bid by the Court.

DEPOSIT

All deposits shall be held by the Receiver in a single non-interest bearing account designated solely for such purpose. A deposit paid by a Successful Bidder shall be dealt with in accordance with these SISP Procedures and the definitive agreement(s) for the transactions contemplated by the Successful Bid. Deposits paid by Qualified Bidders not selected as the Successful Bidder shall be returned to such Qualified Bidder within five business days of Court approval of the Successful Bid.

AMENDMENTS

Any amendments to this SISP may only be made by the Receiver (acting reasonably), or by further order of the Court.

CONFIDENTIALITY AND ACCESS TO INFORMATION

All discussions regarding a proposed transaction, Sale Proposal, Investment Proposal, Bid or Successful Bid should be directed through the Receiver. Under no circumstances should the management of SolvAQUA or its customers be contacted directly without the prior consent of the Receiver. Any such unauthorized contact or communication could result in exclusion of the interested party from the SISP at the discretion of the Receiver.

Participants and prospective participants in the SISP shall not be permitted to receive any information that is not made generally available to all participants relating to the number or identity of Qualified Bidders, Bids, Qualified Bids, the details of any Bids or Qualified Bids submitted or the details of any confidential discussions or correspondence between the Receiver and such other Qualified Bidders in connection with the SISP, except to the extent that the Receiver, with the consent of the applicable participants, is seeking to combine separate Bids into a single Bid.

Notwithstanding the foregoing, under no circumstances will the Receiver share any material information concerning any of the Bids with any person, except as provided for in the Auction Procedure.

SUPERVISION OF THE SISP

This SISP does not, and will not be interpreted to, create any contractual or other legal relationship between the Receiver and any Qualified Bidder, or any other party, other than as specifically set forth in a definitive agreement that may be signed with the Receiver.

Without limiting the preceding paragraph, the Receiver shall not have any liability whatsoever to any person or party, including without limitation any Qualified Bidder, the Successful Bidder, the Company or any other creditor or other stakeholder of the Company, for any act or omission related to the process contemplated by this SISP Procedure, except to the extent such act or omission is the result of gross negligence or willful misconduct of the Receiver. By submitting a Bid, each Qualified Bidder or Successful Bidder shall be deemed to have agreed that it has no claim against the Receiver in respect of the SISP for any reason whatsoever, except to the extent that such claim is the result of gross negligence or willful misconduct by the Receiver.

Participants in the SISP are responsible for all costs, expenses and liabilities incurred by them in connection with the submission of any Bid, due diligence activities, and any further negotiations or other actions whether or not they lead to the consummation of a transaction.

Schedule "A"

Addresses of Receiver

To the Receiver

MNP Ltd.

	algary, AB	T2P 3G4
At	tention:	Vanessa Allen
Er	mail:	vanessa.allen@mnp.ca

Schedule "B"

AUCTION PROCEDURES

- <u>Auction</u>: If the Receiver receives at least one Qualified Bid (other than the Stalking Horse Agreement), the Receiver will conduct and administer the Auction on or before November 30, 2022 at 10:00 a.m. (Mountain Time) via video conference, in accordance with the terms of the SISP.
- 2. Participation: Not less than 48 hours prior to the Auction, the Receiver will:
 - a. advise those Qualified Bidders that provided a Qualified Bid by the Bid Deadline, including the Stalking Horse Bidder (collectively, the "Qualified Parties") that their Qualified Bid will be advanced to the Auction and they shall be eligible to participate in the Auction; and
 - b. provide the Qualified Parties with instructions to participate in the Auction.

Each Qualified Party must inform the Receiver no later than 24 hours prior to the Auction whether it intends to participate in the Auction. If no Qualified Party provides such expression of intent, the Stalking Horse Agreement shall be the Successful Bid.

- <u>Starting Bid.</u> Not less than 24 hours prior to the Auction, the Receiver will advise the Qualified Parties which of the Qualified Bids the Receiver has determined in its reasonable business judgment, after consultation with its advisors, constitutes the then highest or otherwise best Bid (the "Starting Bid").
- 4. Auction Procedures: The Auction shall be governed by the following procedures:
 - a. <u>Attendance</u>. Unless otherwise ordered by the Court, only the authorized representatives, professional advisors or agents of the Receiver and each Qualified Party shall be eligible to attend the Auction and only the authorized representative of each Qualified Party shall be eligible to make any Subsequent Bids (as defined below) at the Auction. All Qualified Parties participating in the Auction must have at least one individual representative with authority to bind the Qualified Parties present at the Auction.
 - <u>Recording</u>. The Auction shall be recorded by the Receiver for their exclusive use and shall not be recorded by any other party.
 - c. <u>Bidding Disclosure</u>. At the Auction, all Qualified Parties shall be permitted to increase their Qualified Bids in accordance with the procedures set forth herein (each, a "Subsequent Bid"). All Subsequent Bids presented during the Auction shall be made and received via video conference on an open basis. All participating Qualified Parties shall be entitled to be present for all bidding with the understanding that the true identity of each participating Qualified Party shall be fully disclosed to all other Qualified Parties and that all material terms of each Subsequent Bid presented during the Auction will be fully disclosed to the Qualified Parties throughout the entire Auction.
 - d. <u>Bidding Commencement</u>. Bidding at the Auction will begin with the Starting Bid and continue, in one or more rounds of bidding, so long as during each round at least one Subsequent Bid is submitted by a Qualified Party that (i) improves upon the Starting Bid or such Qualified Party's immediately prior Subsequent Bid, as the case may be, (ii) proceeds in a minimum additional cash

increment of \$50,000, and (iii) the Receiver determines, in its reasonable business judgment, after consultation with its advisors, such Subsequent Bid is a higher or otherwise better offer than the then current leading Starting Bid or Subsequent Bid.

- e. <u>Evaluation of Subsequent Bids</u>. For the purpose of evaluating the value of the consideration provided by each Subsequent Bid presented at the Auction, the value will: (i) be deemed to be the net consideration payable to the Receiver after considering, *inter alia*, any Expense Reimbursement and Transaction Fee due to the Stalking Horse Bidder under the Stalking Horse Agreement; and (ii) take into account any additional liabilities to be assumed by the Qualified Bidder.
- f. <u>Bidding Rounds</u>. After the first round of bidding and between each subsequent round of bidding, the Receiver shall announce the Subsequent Bid that the Receiver has determined in its reasonable business judgment, after consultation with its advisors, to be the then highest or best bid (the "Leading Bid"). A round of bidding will conclude after each participating Qualified Party has had an opportunity to submit a Subsequent Bid with full knowledge of the Leading Bid.
- g. <u>Bidding Conclusion</u>. The Auction shall continue in one of more rounds and will conclude 30 minutes after each participating Qualified Party has had the opportunity to submit one or more Subsequent Bids with full knowledge of the Leading Bid but where no Qualified Party chooses to submit a Subsequent Bid or where any Subsequent Bid submitted is deemed not to be superior to the Leading Bid as determined by the Receiver acting in its sole discretion.
- h. No Post-Auction Bids. No bids will be considered for any purpose after the Auction has concluded.
- 5. Selection of Successful Bid. Before the conclusion of the Auction, the Receiver will:
 - a. review each Subsequent Bid considering those factors set out in the SISP, 4(e) of these Auction Procedures and, among other things, (i) the amount of consideration being offered and, if applicable, the proposed form, composition and allocation of same, (ii) the value of any assumption of liabilities or waiver of liabilities not otherwise accounted for in subsection (i) above; (iii) the likelihood of the Qualified Party's ability to close a transaction by December 21, 2022 (including factors such as the transaction structure and execution risk), (iv) the likelihood of the Court's approval of the Successful Bid, (v) the net benefit to the receivership estate, and (vi) any other factors the Receiver may reasonably deem relevant; and
 - b. identify the highest or otherwise best bid received at the Auction (the "Successful Bid" and the Qualified Party making such bid, the "Successful Party"), whereupon the Auction will be concluded.
- 6. <u>Definitive Agreement</u>. The Successful Party shall complete and execute binding agreements, contracts, instruments or other documents (collectively, the "**Definitive Agreements**") evidencing and containing the terms and conditions upon which the Successful Bid was made within one (1) business day of the Successful Bid being selected as such, unless extended by the Receiver in its sole discretion. The Definitive

Agreements shall be on substantially the same terms are the Successful Party's Qualified Bid, as modified by the terms of the Successful Bid.

7. <u>Additional Rules</u>. At the Auction, the Receiver, after consultation with its advisors, may employ and announce additional procedural rules that are fair and reasonable under the circumstances (e.g., the amount of time allotted to make Subsequent Bids) for conducting the Auction; provided, however, that such rules are (a) not inconsistent with the Auction Procedures, the *Bankruptcy and Insolvency Act (Canada)*, any order of the Courts entered in connection with the Auction Procedures, and (b) disclosed to each Qualified Party at the Auction.

DocuSign Envelope ID: 8EF04403-C733-41BB-9C79-7125459F8C51

Schedule 1.1(jjj) Form of SISP Order

Clerk's Stamp:

COURT FILE NUMBER

2201-09319

COURT

COURT OF KING'S BENCH OF ALBERTA

JUDICIAL CENTRE CALGARY

PLAINTIFF ARNAKI LTD.

DEFENDANT SOLVAQUA INC.

DOCUMENT ORDER APPROVING SALE & INVESTOR SOLICITATION PROCESS AND STALKING HORSE BID

ADDRESS FOR SERVICE AND CONTACT INFORMATION OF PARTY FILING THIS DOCUMENT Cassels Brock & Blackwell LLP Bankers Hall West 3810, 888 3rd St SW Calgary, AB T2P 5C5

P: 403 351 2920 E: joliver@cassels.com / dmarechal@cassels.com

Attention: Jeffrey Oliver / Danielle Marechal

File No: 49076-20

DATE ON WHICH ORDER WAS PRONOUNCED:October 20, 2022LOCATION WHERE ORDER WAS PRONOUNCED:Calgary, AB (via WebEx)NAME OF JUSTICE WHO MADE THIS ORDER:The Honourable Justice B.B. Johnston

UPON THE APPLICATION of MNP Ltd. ("**MNP**") in its capacity as the Court-appointed receiver and manager (in such capacity, the "**Receiver**") of the current and future undertakings, property and assets of SolvAQUA Inc. (the "**Debtor**") for an order approving, among other things: (i) the Receiver's proposed sale and investor solicitation process (the "**SISP**"); and (ii) approving the stalking horse agreement between the Receiver and 2464525 Alberta Ltd. (the "**Stalking Horse Purchaser**") and the Receiver (the "**Stalking Horse Agreement**"), attached as Schedule 2 to the First Report of the Receiver dated October 11, 2022 (the "**First Report**");

AND UPON HAVING READ the order of the Honourable Justice R.A. Neufeld pronounced August 19, 2022, appointing MNP as Receiver (the "**Receivership Order**"), the First Report and the Affidavit of Service;

AND UPON HEARING the submissions of counsel to the Receiver, counsel to Arnaki Ltd. and the Stalking Horse Purchaser and any other interest parties in attendance;

IT IS HEREBY ORDERED AND DECLARED THAT:

 All capitalized terms not otherwise defined in this order (the "Order") shall have the meaning ascribed to them in the SISP attached as Schedule 1.1(iii) to the Stalking Horse Agreement attached as <u>Schedule "A"</u> hereto.

SERVICE

2. Service of notice of this application (the "**Application**") and materials in support of this Order is hereby declared to be good and sufficient, no other person is required to have been served with notice of the Application and time for service of the Application is abridged to that actually given.

APPROVAL OF RECEIVER'S ACTIVITIES

3. The Receiver's actions, conduct and activities, as reported in the First Report, are commercially reasonable and are approved.

APPROVAL OF THE STALKING HORSE AGREEMENT

- 4. The Stalking Horse Agreement, substantially in the form attached as Schedule "A" hereto, is hereby approved and execution of the Stalking Horse Agreement by the Receiver is hereby authorized and approved, and the Receiver is authorized and directed to take such additional steps and execute such additional documents and make such minor amendments to the Stalking Horse Agreement as may be necessary or desirable for the completion of the terms of the Stalking Horse Agreement, in all cases subject to the terms of this Order.
- 5. The Expense Reimbursement and Transaction Fee (as defined in the Stalking Horse Agreement) is hereby approved and the Receiver is authorized and directed to pay the Expense Reimbursement and Transaction Fee in the manner and circumstances descried in the Stalking Horse Agreement and the SISP, as applicable.

APPROVAL OF THE SALE AND INVESTOR SILICITATION PROCESS

6. The SISP is commercially reasonable and is hereby ratified and approved. The Receiver is authorized and directed to implement the SISP and do all things as are reasonably necessary to conduct and give effect to the SISP, and to take such additional steps and execute such additional documents, and make such minor amendments to the SISP as may be necessary or desirable and

LEGAL*56878664.5

not prejudicial to any stakeholder, for the completion of the terms of the SISP. Further, the Receiver is authorized to enter into any resulting agreement(s) or transaction(s), which may arise in connection with the SISP, as the Receiver determines is necessary or advisable.

- 7. Nothing herein shall act as authorization or approval of the transfer or vesting of any or all of the Debtor's property, assets or undertakings under any sale agreement, or otherwise. Such transfer and vesting shall be dealt with if and when a Successful Bid is selected and shall be subject to further order of this Honourable Court.
- 8. The Receiver is authorized to apply to this Honourable Court:
 - (a) to amend, vary or seek any advice, assistance or direction with respect to the SISP and Stalking Horse Agreement, as may be necessary in order to give full effect to the terms of this Order or as otherwise desirable in the Receiver's discretion; and
 - (b) for an order vesting title to the purchased assets in favour of the Successful Bidder in accordance with, and as defined in, the SISP.
- 9. The Receiver and its affiliates, partners, directors, employees, agents and controlling persons shall have no liability with respect to any and all losses, claims, damages or liabilities, of any nature or kind, to any person in connection with or as a result of the SISP, except to the extent such losses, claims, damages or liabilities result from the gross negligence or willful misconduct of the Receiver in performing its obligations under the SISP.
- 10. In connection with the SISP and pursuant to section 7(3)(c) of the Personal Information Protection and Electronic Documents Act (Canada), the Receiver is authorized and permitted to disclose personal information of identifiable individuals to prospective purchasers or offerors and to their advisors, but only to the extent desirable or required to negotiate and attempt to complete one or more transactions (each, a "Transaction"), in keeping with the SISP. Each prospective purchaser or offeror to whom such information is disclosed shall maintain and protect the privacy of such information and shall limit the use of such information to its evaluation of the Transaction, and if it does not complete a Transaction, shall: (i) return all such information to the Receiver; (ii) destroy all such information to the extent it is reasonably practical to do so. The purchaser of any Property, as defined in the SISP, shall be entitled to continue to use the personal information provided to it, and related to the Property purchased, in a manner which is in all material respects identical to the prior use of such information by the Debtor, and shall return all other personal information to the Receiver, or ensure that other personal information is destroyed.

FOREIGN RECOGNITION

11. This Honourable Court hereby requests the aid and recognition of any court, tribunal, regulatory or administrative body having jurisdiction in Canada or in any of its provinces or territories or in any foreign jurisdiction, to act in aid of and to be complimentary to this Court in carrying out the terms of this Order, to give effect to this Order and to assist the Receiver and its agents in carrying out the terms of this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such order and to provide such assistance to the Receiver, as an officer of the Court, as may be necessary or desirable to give effect to this Order or to assist the Receiver and its agents in carrying out the terms of this Order.

SERVICE

- 12. Service of this Order shall be deemed good and sufficient by:
 - (a) serving the same on the persons listed on the service list created for this proceeding, as amended; and
 - (b) posting a copy of this Order on the Receiver's website at: https://mnpdebt.ca/en/corporate/corporate-engagements/SolvAQUA-inc

and service on any other person is hereby dispensed with.

Service of this Order may be effected by facsimile, electronic mail, personal delivery or courier.
Service is deemed to be effected the next business day following transmission or delivery of this Order.

Justice of the Court of King's Bench of Alberta

SCHEDULE "A"

Stalking Horse Agreement

LEGAL*56878664.5

THIS IS EXHIBIT "B' TO THE AFFIDAVIT OF TIFFANY NESS

SWORN ON THIS 19TH DAY OF OCTOBER 2022 A Commissioner for Oaths in and for Alberta JODIE Kosman

JODIE Kosman A Commissioner for Oaths In and for Alberta My Commission Expires JUNES 2024

ASSET PURCHASE AGREEMENT

MNP LTD., solely in its capacity as receiver and manager of SOLVAQUA INC. and not in its personal capacity

as Seller

- and -

2464525 ALBERTA LTD.

as Buyer

October [+],<u>17,</u> 2022

00609155-1 LEGAL*56914030.6<u>56914030.9</u>

PURCHASE AGREEMENT

THIS AGREEMENT is made as of October [+],17, 2022

BETWEEN:

MNP LTD. ("**MNP**"), solely in its capacity as receiver and manager of the assets, undertakings and properties of SOLVAQUA INC. ("**SolvAQUA**" or the "**Debtor**"), a corporation formed under the laws of the Province of Alberta, and not in its personal capacity

(the "Seller" or the "Receiver")

- and -

2464525 ALBERTA LTD., a corporation incorporated under the laws of the Province of Alberta

(and any of its nominees, designees or assignees hereunder, as applicable, the "Buyer")

RECITALS:

- A. The Debtor carries on the business of providing proprietary wastewater management solutions, based on nanopolymerization technology, and implementing that technology in foreign jurisdictions (together with any other business in which the Debtor is engaged on the date hereof, the "**Business**").
- B. Pursuant to the order of the Honourable Justice Neufeld of the Alberta Court of Queen's Bench (the "Court"), pronounced August 19, 2022 (the "Receivership Order") in Court File No. 2201-09319 (the "Receivership Proceedings"), MNP was appointed as the receiver and manager of all of the assets, undertakings and properties of the Debtor (the "Property").
- C. Pursuant to the Receivership Order, the Receiver is authorized to:
 - (a) market any or all of the Property and negotiate such terms and conditions of sale as the Receiver in its discretion may deem appropriate; and
 - (b) sell, convey, transfer, lease or assign the Property or any part or parts thereof out of the ordinary course of business with the approval of the Court in respect of any transaction in which the purchase price exceeds \$50,000.
- D. The Receiver: (i) wishes to sell, and the Buyer wishes to purchase (and has agreed to act as a "stalking horse bidder" pursuant to the SISP (as defined herein) substantially all the assets and liabilities of the Business; and (ii) intends to seek a Court order approving this Agreement and the SISP.
- E. The Receiver has determined that it is in the best interests of the Debtor's stakeholders to enter into this Agreement and to consummate the transactions contemplated herein on the terms set forth herein.

NOW THEREFORE in consideration of the mutual covenants and agreements contained in this Agreement and other good and valuable consideration (the receipt and sufficiency of which are acknowledged), the Parties agree as follows:

ARTICLE 1 INTERPRETATION

1.1 Definitions

In this Agreement,

- (a) **"Adjustment Date**" has the meaning given to such term in Section 3.6;
- (b) "affiliate" of any Person means, at the time such determination is being made, any other Person controlling, controlled by or under common control with such first Person, in each case, whether directly or indirectly through one or more intermediaries, and "control" and any derivation thereof means the control by one Person of another Person in accordance with the following: a Person ("A") controls another Person ("B") where A has the power to determine the management and policies of B by contract or status (for example the status of A being the general partner of B) or by virtue of beneficial ownership of a majority of the voting interests in B; and for certainty and without limitation, if A owns shares to which are attached more than 50% of the votes permitted to be cast in the election of directors (or other Persons performing a similar role) of B, then A controls B for this purpose;
- (c) "Agreement" means this Asset Purchase Agreement and all attached Schedules, in each case as the same may be supplemented, amended, restated or replaced from time to time, and the expressions "hereof", "herein", "hereto", "hereunder", "hereby" and similar expressions refer to this Agreement and all attached Schedules and unless otherwise indicated, references to Articles, Sections and Schedules are to Articles, Sections and Schedules in this Agreement;
- (d) "Applicable Law" means any domestic or foreign statute, law (including the common law), ordinance, rule, regulation, restriction, by-law (zoning or otherwise), order, or any consent, exemption, approval or license of any Governmental Authority, that applies in whole or in part to the Transaction, the Seller, the Debtors, the Buyer, the Business or any of the Purchased Assets;
- (e) **"Arnaki**" means Arnaki Ltd., a corporation incorporated under the laws of the British Virgin Islands;
- (f) **"Arnaki Loan**" means the loans advanced to the Debtor pursuant to the Arnaki Loan Agreements;
- (g) **"Arnaki Loan Agreements**" means the following loan agreements (as amended and restated) among Arnaki and the Debtor:
 - (i) loan agreement dated August 31, 2020;
 - (ii) loan agreement dated January 4, 2021; and
 - (iii) loan agreement dated February 1, 2021;
- (h) **"Arnaki Loan Contracts**" means the following:
 - (i) Arnaki Loan Agreements;
 - (ii) Arnaki Security;

- (iii) Assignment of Receivables Agreement dated February 1, 2021 between the Debtor and Arnaki; and
- (iv) Extension Agreement dated April 1, 2021 between the Debtor and Arnaki;
- (i) **"Arnaki Security**" means the following security agreements among Arnaki and the Debtor entered into pursuant to the Arnaki Credit Agreements:
 - (i) general security agreement dated August 31, 2020;
 - (ii) general security agreement dated January 4, 2021; and
 - (iii) general security agreement dated February 1, 2021;
- (j) **"Assumed Liabilities**" means those Liabilities incurred after Closing and relating to, or arising in connection with, the Purchased Assets, provided that such Liabilities continue to attach to the Purchased Assets following the granting of the Vesting Order;
- (k) "Business" has the meaning given to such term in Recital A;
- (I) "Business Day" means any day, other than a Saturday or Sunday, on which the principal commercial banks in Calgary are open for commercial banking business during normal banking hours;
- (m) **"Buyer**" has the meaning given to such term in the preamble to this Agreement, or its nominee;
- (n) "Cash Component" means an amount to be determined by the Parties, acting reasonably, provided that such amount is equal to or greater than the sum of the Priority Payables and the Holdback Amount and may include the Vivakor Equipment Purchase Price if it is determined that (i) the Vivakor Equipment is owned by a Person other than SolvAQUA, and (ii) the Buyer elects to include the Vivakor Equipment in the Purchased Assets;
- "Claim" means any right, claim, cause of action or complaint of any Person that may be (0) asserted or made in whole or in part against the Debtor or Seller, any of their respective affiliates and their respective Representatives, whether or not asserted or made in connection with any indebtedness, liability or obligation of any kind whatsoever, and any interest accrued thereon or costs payable in respect thereof, including by reason of the commission of a tort (intentional or unintentional), by reason of any breach of contract or other agreement (oral or written), by reason of any breach of duty (including, without limitation, any legal, statutory, equitable or fiduciary duty) or by reason of any right of ownership of or title to property or assets or right to a trust or deemed trust (statutory, express, implied, resulting, constructive or otherwise), and whether or not any indebtedness, liability or obligation is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured, unsecured, present or future, known or unknown, by guarantee, surety or otherwise, and whether or not any right, claim, cause of action or complaint is executory or anticipatory in nature;
- (p) "Closing" means the completion of the Transaction at the Closing Time;
- (q) **"Closing Date**" means December 21, 2022, or such other date as the Parties may agree, acting reasonably;

- (r) **"Closing Documents**" means all contracts, agreements and instruments required by this Agreement to be delivered at or before the Closing;
- (s) **"Closing Time**" means 10:00 a.m. (Calgary time) on the Closing Date or such other time on the Closing Date as the Parties agree in writing that the Closing Time shall take place;
- (t) "Confidential Information" means non-public, confidential, personal or proprietary information which is furnished to a Party by the other Party, including, without limitation, information about identifiable individuals, any information relating to a Party and its affiliates or any customer or supplier of a Party; provided that "Confidential Information" does not include information that:
 - (i) is or becomes generally available to the public other than as a result of disclosure in breach of this Agreement;
 - (ii) is received by a Party from a third party that obtained it lawfully and was under no duty of confidentiality;
 - (iii) was lawfully in a Party's possession prior to disclosure thereof by the other Party; or
 - (iv) was independently developed by a Party without use of, or reference to, the other Party's Confidential Information;
- "Contracts" means contracts, licences, permits, leases, agreements, commitments, entitlements or engagements to which the Debtor is party or by which the Debtor is bound;
- (v) "Court" has the meaning given to such term in Recital B;
- (w) **"Court Approval**" means the pronouncement of the SISP Order and the Vesting Order by the Court;
- (x) **"CRA Priority Payments**" means those deemed trust amounts validly owing by the Debtor to Canada Revenue Agency under the *Income Tax Act* (Canada) or the *Excise Tax Act* (Canada), or other similar provincial legislation as of the Closing Date that are in priority to the Arnaki Security;
- (y) **"Credit Bid Component**" has the meaning given to such term in Section 3.1;
- (z) "Debtor" has the meaning given to such term in the preamble to this Agreement;
- (aa) "Deposit" means the amount of \$100,000, comprising part of the Purchase Price;
- (bb) **"Employee Priority Payments**" means all amounts payable by the Debtor's estate to current or former employees of the Debtor under Sections 81.4 and 81.6 of the *Bankruptcy and Insolvency Act* (Canada) that rank in priority to the Arnaki Security;
- (cc) "Encumbrances" means all mortgages, pledges, charges, liens, hypothecs, hypothecations, debentures, trust deeds, Claims, assignments by way of security or otherwise, security interests, Priority Charges, conditional sales contracts or other title retention agreements, security created under the *Bank Act* (Canada), rights of first refusal, reservations of ownership or similar interests or instruments charging or creating a security interest in the Purchased Assets or any part thereof or interest therein, and any agreements, leases, licenses, occupancy agreements, options, easements, rights of way,

restrictions, executions, or other encumbrances (including notices or other registrations in respect of any of the foregoing) affecting title to the Purchased Assets or any part thereof or interest therein;

- (dd) "Equipment Owner" means the legal owner of the Vivakor Equipment;
- (ee) **"Excluded Assets**" means the Purchase Price;
- (ff) **"Expense Reimbursement and Transaction Fee**" has the meaning given to such term in Section 9.3(a);
- (gg) "Final" with respect to any order of any court of competent jurisdiction, means that leave to appeal or reconsideration shall not have been sought in respect of such order and that such order shall not have been stayed, appealed, varied (except with the consent of the Buyer and Seller) or vacated, and all time periods within which leave to appeal and reconsideration could at law be sought shall have expired or been waived and all time periods within which such order could at law be appealed shall have expired;
- (hh) "GST" means the sales tax payable under the GST Legislation;
- (ii) "GST Legislation" means Part IX of the Excise Tax Act (Canada);
- (jj) **"Governmental Authority**" means any government, regulatory authority, governmental department, agency, commission, bureau, court, judicial body, arbitral body or other law, rule or regulation-making entity:
 - having jurisdiction over the Debtor, the Buyer or the Purchased Assets on behalf of any country, province, state, locality or other geographical or political subdivision thereof; or
 - (ii) exercising or entitled to exercise any administrative, judicial, legislative, regulatory or taxing authority or power;
- (kk) "Governmental Authorizations" means the permits, licences, approvals and authorizations, orders, certificates, consents, directives, notices, licences, permits, variances, registrations or other rights issued to or held or required by the Seller and/or the Debtors relating to the Business or any of the Purchased Assets by or from any Governmental Authority;
- (II) **"Holdback Amount**" means the amount of \$75,000 paid by the Buyer to the Seller to account for any current or future Priority Payables, the amount of which Priority Payables cannot be definitively ascertained as at the Closing Date;
- (mm) **"including**" and **"includes**" shall be interpreted on an inclusive basis and shall be deemed to be followed by the words "without limitation";
- (nn) "Indebtedness" means the aggregate of all debts, liabilities and obligations owing to the Buyer by the Debtor, including without limitation the amounts outstanding under the Arnaki Loan, inclusive of all interest, costs, fees, expenses and amounts that may accrue in accordance with the Arnaki Loan and Arnaki Loan Contracts;
- (oo) **"Intellectual Property**" means any, and all, of the following in any jurisdiction throughout the world:

- (i) trademarks, including all applications and registrations and the goodwill connected with the use and symbolized by the foregoing;
- (ii) copyrights and industrial designs, including all applications and registrations relating to the foregoing;
- (iii) trade secrets and confidential know-how;
- (iv) patents and patent applications;
- (v) websites and internet domain name registrations; and
- (vi) other intellectual property and related proprietary rights, interests and protections.
- (pp) "Liabilities" means liabilities, obligations or commitments of any nature whatsoever, whether asserted or unasserted, known or unknown, absolute or contingent, accrued or unaccrued, matured or unmatured or otherwise.
- "Material Adverse Change" or "Material Adverse Effect" means any change, (qq) development, effect, event, circumstance, fact or occurrence that, individually or in the aggregate with such other changes, developments, effects, events, circumstances, facts or occurrences, is, or would reasonably be expected to be, material and adverse to the properties, assets, liabilities (contingent or otherwise), condition (financial or otherwise), operations or results of operations of the Business; other than any change, development, effect, event, circumstance, fact or occurrence arising out of, attributable to or resulting from: (A) any action expressly required or permitted by this Agreement or relating to the Debtor's current financial condition, including the Receivership Proceedings; (B) general political, economic or financial conditions in Canada or elsewhere in the world; (C) any change generally affecting the industries in which the Business is conducted (including changes in general market prices, or regulatory changes in any such industry); (D) acts of terrorism or war (whether or not declared); (E) any changes to existing Applicable Law (including the interpretation thereof); (F) any changes to IFRS or the adoption, implementation or proposal of any new accounting principles; (G) hurricanes, earthquakes, storms, floods or other natural disasters, epidemics, pandemics (including COVID), outbreak or escalation of hostilities, the declaration of war, acts of terrorism, or acts of God; (H) any action consented to by the Buyer; (I) any failure by the Debtor to meet any projections or estimates (including internal projections or estimates) of revenues, earnings, working capital or performance for any period; or (J) any action, change, development, effect, event, circumstance, fact or occurrence that is attributable to or otherwise caused by the Seller, the Debtor or the Buyer;
- (rr) "Parties" means the Seller and the Buyer collectively, and "Party" means either the Seller or the Buyer; provided, however that, Parties and Party may also refer to the Debtors, as applicable;
- (ss) **"Permitted Encumbrances**" means those permitted encumbrances (if any) provided for in the Vesting Order;
- (tt) "Person" means any individual, partnership, limited partnership, limited liability company, joint venture, syndicate, sole proprietorship, company or corporation with or without share capital, unincorporated association, trust, trustee, executor, administrator or other legal personal representative, Governmental Authority or other entity however designated or constituted;

- (uu) **"Priority Charges**" means any amounts that rank in priority to the Arnaki Security, including without limitation any Court ordered charges or statutory priority claims (whether arising in the Receivership Proceeding or any other proceeding involving the Debtor, including without limitation a bankruptcy proceeding), the Receivership Costs, the Employee Priority Payments and the CRA Priority Payments;
- (vv) **"Priority Payables**" means any current or future amounts owing as secured by the Priority Charges;
- (ww) "Purchase Price" has the meaning given to such term in Section 3.1;
- (xx) **"Purchased Assets**" means those assets, undertakings and properties listed in Schedule 1.1(xx), but for greater certainty, shall not include the Excluded Assets;
- (yy) "Receiver" has the meaning given to such term in Recital B;
- (zz) **"Receiver's Borrowing Charge**" has the meaning given to such term in the Receivership Order;
- (aaa) **"Receiver's Certificate**" means the certificate filed with the Court by the Receiver certifying that the Receiver is satisfied that: (i) all conditions to Closing set forth in Article 6 have been satisfied or waived; and (ii) the Buyer has paid, and the Seller has received, the Purchase Price;
- (bbb) "Receiver's Charge" has the meaning given to such term in the Receivership Order;
- (ccc) "**Receivership Costs**" means all of the taxable fees, costs and expenses associated with the Receiver's administration of the Debtor's estate pursuant to the Receivership Order, including without limitation, the Receiver's Fees, Receiver's Charge and Receiver's Borrowing Charge.
- (ddd) "**Receiver's Fees**" means the fees and disbursements of the Receiver and the Receiver's counsel incurred in connection with the receivership of the Debtor, secured by the Receiver's Charge, as approved by the Court from time to time;
- (eee) "Receivership Order" has the meaning given to such term in Recital B;
- (fff) "Receivership Proceedings" has the meaning given to such term in Recital B;
- (ggg) **"Representative"** means, in respect of a Person, each director, officer, employee, agent, legal counsel, accountant, consultant, contractor, professional advisor and other representative of such Person and its affiliates;
- (hhh) "Seller" has the meaning given to such term in the preamble to this Agreement;
- (iii) **"SISP**" means the sale and investment solicitation process set forth in Schedule "1.1(iii)" hereto;
- (jjj) "SISP Order" means an order of the Court substantially in the form attached as Schedule "1.1(jjj)" hereto that, among other things, approves the SISP and related matters, with any amendments thereto to be acceptable to each of the Seller and the Buyer, each acting reasonably.

- (kkk) "SolvAQUA Claims" means any and all claims of the Debtor, including but not limited to any claims related to alleged breaches under various contracts between SolvAQUA and each of Vivaventures Inc. and Clear Capital Partners as well as any corresponding insurance claims related to coverage provided by Export Development Canada in relation to these contracts;
- (III) "Successful Bid" has the meaning ascribed to it in the SISP;

(mmm) "Tax" and "Taxes" means any and all:

- taxes, duties, fees, excises, premiums, assessments, imposts, levies and other charges or assessments of any kind whatsoever imposed by any Governmental Authority, including those with respect to income, goods and services, harmonized sales, transfer, land transfer, use, real or personal property, and registration fees; and
- (ii) all interest, penalties, fines, additions to tax or other additional amounts imposed by any Governmental Authority on or in respect of amounts of the type described in clause (i) above or this clause (ii);
- (nnn) "Tax Act" means the Income Tax Act (Canada), as amended;
- (000) **"Transaction**" means, collectively, the sale and purchase of the Purchased Assets pursuant to this Agreement and all other transactions contemplated by this Agreement that are to occur contemporaneously with the sale and purchase of the Purchased Assets;
- (ppp) "Transfer Taxes" has the meaning given to such term in Section 7.7(c);
- (qqq) "Vesting Order" means an order of the Court substantially in the same form as the Alberta Model Sale Approval and Vesting Order, with any amendments thereto to be acceptable to each of Seller and Buyer, each acting reasonably;
- (rrr) **"Vivakor Equipment**" means two completed wastewater management units with serial number J-888011-A; and
- (sss) **"Vivakor Equipment Purchase Price**" means an amount to be determined by the Equipment Owner and the Buyer, acting reasonably, for the purchase of the Vivakor Equipment; provided that it is determined by: (i) the Receiver, Buyer and Equipment Owner; or (ii) the Court, that the Vivakor Equipment is owned by a Person other than SolvAQUA.

1.2 Schedules

The schedules to this Agreement are an integral part of this Agreement.

<u>Schedule</u>	Description
Schedule 1.1(xx)	Purchased Assets
Schedule 1.1(iii)	SISP
Schedule 1.1(jjj)	Form of SISP Order

1.3 Statutes

Unless specified otherwise, reference in this Agreement to a statute refers to that statute as it may be amended, or to any restated or successor legislation of comparable effect.

1.4 Headings and Table of Contents

The inclusion of headings and a table of contents in this Agreement is for convenience of reference only and shall not affect the construction or interpretation hereof.

1.5 Gender and Number

In this Agreement, unless the context otherwise requires, words importing the singular include the plural and vice versa and words importing gender include all genders.

1.6 Currency

Except where otherwise expressly provided, all amounts in this Agreement are stated and shall be paid in Canadian dollars.

1.7 Invalidity of Provisions

Each of the provisions contained in this Agreement is distinct and severable and a declaration of invalidity or unenforceability of any such provision or part thereof by a court of competent jurisdiction shall not affect the validity or enforceability of any other provision hereof.

1.8 Entire Agreement

This Agreement and the agreements and other documents required to be delivered pursuant to this Agreement, constitute the entire agreement between the Parties and set out all the covenants, promises, warranties, representations, conditions and agreements between the Parties in connection with the subject matter of this Agreement and supersede all prior agreements, understandings, negotiations and discussions, whether oral or written, pre-contractual or otherwise. There are no covenants, promises, warranties, representations, conditions, understandings or other agreements, whether oral or written, pre-contractual or otherwise, express, implied or collateral between the Parties in connection with the subject matter of this Agreement except as otherwise specifically set forth in this Agreement and any document required to be delivered pursuant to or in respect of this Agreement.

1.9 Waiver, Amendment

Except as expressly provided in this Agreement, no amendment or waiver of this Agreement shall be binding unless executed in writing by all Parties hereto. No waiver of any provision of this Agreement shall constitute a waiver of any other provision nor shall any waiver of any provision of this Agreement constitute a continuing waiver unless otherwise expressly provided.

1.10 Governing Law; Jurisdiction and Venue

This Agreement, the rights and obligations of the Parties under this Agreement, and any claim or controversy directly or indirectly based upon or arising out of this Agreement or the Transaction (whether based on contract, tort, or any other theory), including all matters of construction, validity and performance, shall in all respects be governed by, and interpreted, construed and determined in accordance with, the laws of the Province of Alberta and the federal laws of Canada applicable therein, without regard to the conflicts of law principles thereof. The Parties consent to the exclusive jurisdiction and venue of the Court prior to a Final order of the Court terminating the Receivership Proceedings and thereafter to the Courts of Alberta for the resolution of any disputes arising under this Agreement. Each

Party agrees that service of process on such Party as provided in Section 11.6 shall be deemed effective service of process on such Party.

ARTICLE 2 PURCHASE AND SALE

2.1 Agreement to Purchase and Sell Purchased Assets

- (a) Upon and subject to the terms and conditions of this Agreement at Closing and effective as of the Closing Time the Seller shall sell to the Buyer, and the Buyer shall purchase from the Seller, free and clear of all Encumbrances, the Purchased Assets.
- (b) Provided that Closing occurs, possession risk, beneficial and legal ownership of the Purchased Assets shall, transfer from the Seller to the Buyer in accordance with the Vesting Order and the terms of this Agreement.

2.2 "As is, Where Is"

The Buyer acknowledges and agrees that all of the Purchased Assets are being purchased on an "as is, where is" basis as they shall exist at Closing. Unless and solely to the extent expressly set forth in this Agreement, no representation, warranty or covenant is expressed or implied by the Seller, including any warranties as to title, encumbrances, description, merchantability or fitness for a particular purpose, environmental compliance, condition, quantity or quality, or in respect of any other matter or thing whatsoever concerning the Business and/or the Purchased Assets or the right of the Seller to sell or assign the same. Without limiting the generality of the foregoing, any and all conditions, warranties, or representations expressed or implied pursuant to the *Sale of Goods Act* (Alberta), as amended, or similar legislation do not apply hereto and have been waived by the Buyer. This Section 2.2 shall not merge on Closing and is deemed incorporated by reference into all Closing Documents and deliveries.

ARTICLE 3 PURCHASE PRICE AND RELATED MATTERS

3.1 Purchase Price

The purchase price payable by the Buyer to the Seller for the Purchased Assets (the "**Purchase Price**") shall be the aggregate sum of:

- (a) a cash payment of the Cash Component; and
- (b) the balance of the Purchase Price, equal to the amount of \$2,500,000, to be paid by way of set off against (as a non-cash credit reduction of) the Indebtedness (the "Credit Bid Component").

3.2 Deposit

- (a) The Deposit shall be paid by wire transfer in immediately available funds within two (2) Business Days of the SISP Order having been granted by the Court. The Deposit shall be held, pending Closing, by the Seller's counsel in a non-interest bearing account.
- (b) If the Closing does not occur solely by reason of the default of the Buyer, the full amount of the Deposit, less any applicable withholding tax, shall become the property of and be retained by the Seller to compensate it for expenses incurred in connection with the Transaction contemplated in this Agreement and the delay caused to the Seller's efforts to sell the Purchased Assets. The entitlement of the Seller to retain the Deposit in such

circumstances shall not limit the Seller's right to exercise any other rights which the Seller may have against the Buyer in respect of such default.

(c) If the Closing does not occur for any reason other than the default of the Buyer, the full amount of the Deposit, less any applicable withholding tax, shall be returned forthwith to the Buyer. It is understood that non-completion of the transaction of purchase and sale contemplated by this Agreement by the Buyer because of the non-fulfillment of any conditions precedent set out in Section 6.1 (where such condition precedent was in favour of the Buyer) or 6.2 shall not be considered to be a default of the Buyer.

3.3 Satisfaction of Purchase Price

The Buyer shall satisfy the Purchase Price at the Closing Time by:

- (a) irrevocably and unconditionally authorizing the Seller's counsel to release the Deposit to the Seller, and the Buyer paying to the Receiver in trust the Cash Component (less the Deposit) by wire transfer in immediately available funds pursuant to wire transfer instructions to be provided by the Receiver to the Buyer no later than one (1) Business Day prior to the Closing Date; and
- (b) delivering an acknowledgement <u>efsigned by</u> the Buyer<u>and Arnaki</u> that the Indebtedness has been reduced by the amount of \$2,500,000 in consideration of the balance of the Purchase Price.

3.4 Purchase Price Allocation

The Purchase Price shall be allocated among the Purchased Assets as agreed by the Parties herein and/or in separate written instruments, acting reasonably. Such allocation shall be binding and the Buyer and the Seller shall report the purchase and sale of the Purchased Assets and file all filings which are necessary or desirable under the Tax Act and any other applicable Tax legislation to give effect to such allocations and shall not take any position or action inconsistent with such allocation.

3.5 Payment of Certain Liabilities

Upon payment of the Cash Component to the Seller, the Seller shall satisfy all known Priority Payables as required to be paid on Closing, in accordance with the Vesting Order.

3.6 Adjustments

The Parties will make, without duplication, the usual adjustments relating to the Purchased Assets, as of the Closing Date.

3.7 Determination of Final Purchase Price

In addition to those adjustments described in Section 3.6, the Parties acknowledge that adjustments will be required after the Closing Date to finally determine the Purchase Price once the final amount of the Priority Payables is known. The Purchase Price shall be finally determined on the date that the Receiver is discharged pursuant to a Discharge Order issued by the Court, or such other date as the parties agree to in writing ("**Adjustment Date**"). On the Adjustment Date, the Parties shall agree on the final Purchase Price and the Holdback Amount shall be distributed by the Seller as follows:

(a) Seller shall pay from the Holdback any Priority Payables not previously satisfied under Section 3.5;

- (b) following the payment of any Priority Payables under Section 3.5 and Section 3.7(a), any remainder of the Holdback Amount shall be reimbursed to the Buyer; and
- (c) in the event the Holdback Amount is not sufficient to satisfy the Priority Payables, the Buyer shall pay to the Seller within two (2) Business Days of being notified of such deficiency, the deficiency by way of certified cheque, bank draft or guaranteed electronic funds.

ARTICLE 4

REPRESENTATIONS AND WARRANTIES BY THE SELLER

The Seller represents and warrants to the Buyer and acknowledges that the Buyer is relying upon the following representations and warranties in connection with its purchase of the Purchased Assets the matters set out below:

4.1 Validly Appointed

The Seller has been validly appointed by the Court as receiver and manager of all of the assets, undertakings and properties of the Debtor acquired for, or used in relation to a business carried on by the Debtor, including all proceeds thereof, pursuant to the Receivership Order.

4.2 Recitals

The Recitals hereto are true and correct as of the date of this Agreement.

4.3 Due Authorization and Enforceability of Obligations

Subject to Court Approval being obtained, the Seller has all necessary power, authority and capacity to enter into this Agreement and to carry out its obligations under this Agreement. This Agreement has been, and at the Closing Time the Closing Documents will be, duly executed and delivered by the Seller in its capacity as Receiver.

4.4 Residence of the Seller

The Seller is not a non-resident of Canada within the meaning of section 116 of the Tax Act.

4.5 No Other Representations, Warranties or Covenants

All of the Purchased Assets are being purchased on an "as is where is" basis. Unless and solely to the extent expressly set forth in this Agreement, no representation, warranty or covenant is expressed or implied by the Seller, including any warranties as to title, encumbrance, description, merchantability or fitness for a particular purpose, environmental compliance, condition, quantity or quality, or in respect of any other matter or thing whatsoever concerning the Business and/or the Purchased Assets or the right of the Seller to sell or assign the same. The disclaimer in this Section 4.5 is made notwithstanding the delivery or disclosure to the Buyer or its directors, officers, employees, agents or representatives of any documentation or other information.

ARTICLE 5 REPRESENTATIONS AND WARRANTIES OF THE BUYER

The Buyer represents and warrants to the Seller as follows, and acknowledges that the Seller is relying upon the following representations and warranties in connection with its sale of the Purchased Assets:

5.1 Existence

The Buyer is duly organized and validly existing under the laws of its jurisdiction of organization.

5.2 Recitals

The Recitals hereto are true and correct as of the date of this Agreement.

5.3 Due Authorization

The Buyer has all necessary power, authority and capacity to enter into this Agreement and to carry out its obligations under this Agreement. The execution and delivery of this Agreement and the consummation of the Transaction have been duly authorized by all necessary action of the Buyer. This Agreement has been duly executed and delivered by the Buyer and constitutes a valid and binding obligation of the Buyer enforceable against it in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, moratorium, reorganization and similar laws affecting creditors generally and by general principles of equity.

5.4 Absence of Conflicts

The Buyer is not a party to, bound or affected by or subject to any provision in its articles, by-laws or other constating documents or Applicable Laws or Governmental Authorizations, approvals, franchises, orders, certificates, consents, directives, notices, licences, permits, variances, registrations or other rights issued, granted or given by or from any Governmental Authority that would be violated, breached by, or under which any default would occur or with notice or the passage of time would, be created as a result of the execution and delivery of, or the performance of obligations under, this Agreement or any other agreement or document to be entered into or delivered under the terms of this Agreement, except for any violations, breaches or defaults or any Applicable Laws or any Governmental Authorizations, approvals, orders, certificates, consents, directives, notices, licences, permits, variances, registrations or other rights issued, granted or given by or from any Governmental Authority, that would not have a material and adverse effect on the ability of the Buyer to consummate the transactions hereunder.

5.5 No Litigation

Other than the Receivership Proceedings, there is no Claim that is pending or, to the Buyer's knowledge, threatened in any court or before any Governmental Authority that would adversely affect the Buyer's ability to perform its obligations under this Agreement on a timely basis.

5.6 Approvals and Consents

Except for Court Approval, no authorization, consent or approval of, or filing with or notice to, any Governmental Authority or any other Person is required in connection with the execution, delivery or performance by the Seller of any of its covenants and obligations under this Agreement and each of the Closing Documents to be executed and delivered by the Seller hereunder, including the purchase of any of the Purchased Assets hereunder.

5.7 GST Registration

The Buyer, or its assignee(s)nominee acquiring the Purchased Assets, is, or at the Closing Time will be, registered for purposes of the GST Legislation and will provide its registration number to the Seller.

5.8 Acquisition of Business

Buyer is acquiring under this Agreement all or substantially all the property that can reasonably be regarded as being necessary for it to carry on the Business as a business.

5.9 Financial Ability

The Buyer has and will have at all relevant times, the financial ability and sufficient funds to perform all of its obligations under this Agreement, and the availability of such funds will not be subject to the consent, approval or authorization of any Person or the availability of any financing.

5.10 No Additional Due Diligence

The Buyer acknowledges and agrees that: (a) it has had an opportunity to conduct any and all due diligence regarding the Purchased Assets and the Business prior to the execution of this Agreement; (b) it has relied solely upon its own independent review, investigation and/or inspection of any documents and/or the Purchased Assets and/or the Business; (c) it is not relying upon any written or oral statements, representations, promises, warranties or guaranties whatsoever, whether express or implied, (by operation of law or otherwise), regarding the Purchased Assets or the Business or the completeness of any information provided in connection therewith, except as expressly stated in this Agreement; and (d) the obligations of the Buyer under this Agreement are not conditional upon any additional due diligence.

ARTICLE 6 CONDITIONS

6.1 Conditions for the Benefit of the Buyer and the Seller

The respective obligations of the Buyer and of the Seller to consummate the Transaction are subject to the satisfaction of, or compliance with, at or prior to the Closing Time, each of the following conditions:

- (a) no provision of any Applicable Law and no judgment, injunction, order or decree that prohibits the consummation of the purchase of the Purchased Assets pursuant to this Agreement shall be in effect;
- (b) the Receivership Order and the SISP Order shall have been issued and entered, and shall be Final, and the Receivership Proceedings shall be ongoing;
- (c) the Stalking Horse Bidder shall be the Successful Bidder (each, as defined in the SISP) under the SISP;
- (d) the Vesting Order shall have been issued and entered on or before November 30, 2022 or on or before such later date as the Parties agree to in writing, and shall be Final.

6.2 Conditions for the Benefit of the Buyer

The obligation of the Buyer to consummate the Transaction is subject to the satisfaction of, or compliance with, or waiver by the Buyer of, at or prior to the Closing Time, each of the following conditions (each of which is acknowledged to be for the exclusive benefit of the Buyer):

(a) the representations and warranties of the Seller set forth in this Agreement shall be true and correct at the Closing Time with the same force and effect as if made at and as of such time, except where any failure or failures of any such representations and warranties to be so true and correct would not, individually or in the aggregate cause a Material Adverse Change (and, for this purpose, any reference to "material", "Material Adverse Change" or any other concept of materiality in such representations and warranties shall be ignored);

- (b) the covenants contained in this Agreement to be performed by the Seller at or prior to the Closing Time shall have been performed in all material respects as at the Closing Time;
- (c) the Buyer shall have received a certificate confirming the satisfaction of the conditions contained in Sections 6.2(a) and 6.2(b), signed for and on behalf of the Seller without personal liability by an authorized signatory of the Seller or other Persons reasonably acceptable to the Buyer, in each case in form and substance reasonably satisfactory to the Buyer;
- (d) the Closing Documents, all other documents relating to the due authorization and completion of the Transaction and all actions and proceedings taken on or prior to the Closing in connection with the performance by the Seller of its obligations under this Agreement shall be satisfactory to the Buyer, acting reasonably, and the Buyer shall have received copies of all such documents and evidence that all such actions and proceedings have been taken as it may reasonably request in form and substance reasonably satisfactory to the Buyer;
- (e) the Purchased Assets shall be assigned and transferred to the Buyer free and clear of all Encumbrances, in accordance with the Vesting Order;
- (f) since the date first written above, no Material Adverse Change shall have occurred.

6.3 Conditions for the Benefit of the Seller

The obligation of the Seller to consummate the Transaction is subject to the satisfaction of, or compliance with, or waiver where applicable, by the Seller of, at or prior to the Closing Time, each of the following conditions (each of which is acknowledged to be for the exclusive benefit of the Seller):

- (a) the representations and warranties of the Buyer set forth in this Agreement shall be true and correct in all material respects at the Closing Time with the same force and effect as if made at and as of such time, except where any failure or failures of any such representations and warranties to be so true and correct would not, individually or in the aggregate cause a Material Adverse Change (and, for this purpose, any reference to "material", "Material Adverse Change" or any other concept of materiality in such representations and warranties shall be ignored);
- (b) the covenants contained in this Agreement to be performed by the Buyer at or prior to the Closing Time, including payment of the Purchase Price, shall have been performed in all material respects as at the Closing Time;
- (c) the Seller shall have received a certificate confirming the satisfaction of the conditions contained in Sections 6.3(a) and 6.3(b) signed for and on behalf of the Buyer without personal liability by an executive officer of the Buyer or other persons reasonably acceptable to the Seller, in each case in form and substance reasonably satisfactory to the Seller;
- (d) all other Closing Documents required pursuant to this Agreement to be delivered by the Buyer on Closing in form and substance reasonably satisfactory to the Seller;
- (e) the Buyer shall have delivered to the Seller the Deposit as required under Section 3.2;

- (f) the Buyer shall have delivered to the Seller all of the deliverables contained in Section 10.2 in form and substance satisfactory to the Seller acting in its sole discretion; and-
- (g) the acknowledgement of the Buyer, Buyer shall have obtained an acknowledgment from Arnaki confirming that the Indebtedness has been reduced by the amount of \$2,500,000 in consideration of the balance of the Purchase Price, and
- (h) the Buyer shall have been delivered and released by the Buyer to the Seller, the acknowledgment required under Section 3.3(b) confirming that the Indebtedness has been reduced by the amount of \$2,500,000 in consideration of the balance of the Purchase Price.

ARTICLE 7 ADDITIONAL AGREEMENTS OF THE PARTIES

7.1 Access to Information

Until the Closing Time, the Seller and/or the Debtor shall furnish the Buyer's personnel engaged in the Transaction and their accountants, legal advisers, consultants and representatives during normal business hours, all such information relating to the Purchased Assets as the Buyer may reasonably request in connection with the Transaction. Notwithstanding the foregoing, the Seller and/or the Debtor shall not be required to disclose any information, records, files or other data to the Buyer where prohibited by any Applicable Laws or such disclosure would have the effect of causing the waiver of any solicitor-client privilege.

7.2 Conduct of Business Until Closing Time

Except: (1) as expressly required by this Agreement; and (2) with the prior written consent of the Buyer (not to be unreasonably withheld, conditioned or delayed), as necessary or advisable in connection with or pursuant to the Receivership Proceedings, or as required by Applicable Law (including to the extent required by any Contract), to the extent reasonably necessary, and as practicable having regard to the Receivership Proceedings and any Court order issued in the Receivership Proceedings, the Seller shall:

- (a) not amend, terminate or assign any Contracts that are included in the Purchased Assets and material to the Business;
- (b) not waive or assign any material rights of the Business under any Contract included in the Purchased Assets and material to the Business; and
- (c) not enter into any lease, contract or agreement, licence or other commitment related to the Business that would constitute a personal property lease, real property lease or Contract except, in each case, in the ordinary course of the Business.

7.3 Further Assurances

Each of the Parties hereto shall promptly do, make, execute or deliver, or cause to be done, made, executed or delivered, all such further acts, documents and things as the other Parties hereto may reasonably require from time to time for the purpose of giving effect to this Agreement and shall use commercially reasonable efforts and take all such steps as may be reasonably within its power to implement to their full extent the provisions of this Agreement. Upon and subject to the terms and conditions of this Agreement and subject to the directions of any applicable courts to the Seller, the Parties shall use their commercially reasonable efforts to take or cause to be taken all actions and to do or cause to be done all things necessary proper or advisable under Applicable Laws to consummate and

make effective the Transaction, including using commercially reasonable efforts to satisfy the conditions precedent to the obligations of the Parties hereto.

7.4 Tax Matters

- (a) The Buyer and the Seller agree to use commercially reasonable efforts to furnish or cause to be furnished to each other, as promptly as practicable, such information and assistance relating to the Purchased Assets as is reasonably necessary for the preparation and filing of any Tax return, claim for refund or other required or optional filings relating to Tax matters, for the preparation for and proof of facts during any Tax audit, for the preparation for any Tax protest, for the prosecution of any suit or other proceedings relating to Tax matters and for the answer to any governmental or regulatory inquiry relating to Tax matters.
- (b) For purposes of any income Tax return related to the Transaction, the Buyer and, to the extent applicable, the Seller, agree to report the Transaction in a manner consistent with the Purchase Price allocation determined in accordance with Section 3.4, and the Buyer and the Seller shall not voluntarily take any action inconsistent therewith in any such Tax return, refund claim, litigation or otherwise, unless required by applicable Tax laws. The Buyer and the Seller shall each be responsible for the preparation of their own statements required to be filed under the Tax Act and other similar forms in accordance with applicable Tax laws.
- (c) All amounts payable by the Buyer to the Seller pursuant to this Agreement are exclusive of any GST, or any other federal, provincial, state or local or foreign value-added, sales, use, consumption, multi-staged, ad valorem, personal property, customs, excise, stamp, transfer, land or real property transfer, or similar Taxes, duties, or charges, or any recording or filing fees or similar charges (collectively, "**Transfer Taxes**"). All Transfer Taxes are the responsibility of and for the account of the Buyer. The Buyer and the Seller agree to cooperate to determine the amount of Transfer Taxes payable in connection with the Transaction. If the Seller is required by Applicable Law or by administration thereof to collect any applicable Transfer Taxes from the Buyer, the Buyer shall pay such amounts to the Seller concurrent with the payment of any consideration payable pursuant to this Agreement, and the Seller shall remit or account for such Transfer Taxes to the applicable Governmental Authority on a timely basis and otherwise in accordance with Applicable Laws.
- (d) The Seller and the Buyer shall jointly make, and the Buyer will file, the election provided for under GST Legislation such that no GST will be payable with respect to the purchase and sale of any of the Purchased Assets. The Buyer shall at all times indemnify and hold harmless the Seller and their directors, officers and employees, against and in respect of any and all amounts assessed by the Minister of National Revenue (Canada) as a consequence of such Minister determining, for any reason, that the election is unavailable, inapplicable, invalid or not properly made.

7.5 Fees and Expenses

Except as expressly provided in this Agreement, all fees and expenses incurred in connection with the negotiation and settlement of this Agreement and the completion of the Transaction, including the fees and disbursements of counsel, financial advisors and accountants, shall be paid by the Person incurring such fees or expenses.

7.6 Advice and Direction

The Parties acknowledge that the Seller is entitled (but not required) to seek the advice and directions of the Court in respect of any determination to be made, consent right to be exercised or other action to be taken by the Seller under this Agreement.

ARTICLE 8 COURT ORDER

8.1 Vesting Order

- (a) The Seller shall file an application with the Court for the issuance of the Vesting Order, which Vesting Order shall only be effective upon the Buyer being determined to be the Successful Bidder and the Receiver filing a certificate substantially in the form attached to the SISP Order. Such application shall be scheduled for a date that is on or before Nevember 30, December 14, 2022, or such later date as may be scheduled upon the agreement of the Parties (acting reasonably), due to court availability or pursuant to direction of the Court.
- (b) The Buyer shall cooperate with the Seller acting reasonably, as may be necessary, in obtaining the Vesting Order.
- (c) The Seller shall use its best efforts to obtain the Vesting Order as promptly as practicable.

ARTICLE 9 TERMINATION

9.1 Termination

This Agreement may be terminated at any time prior to Closing as follows:

- (a) subject to any approvals required from the Court or otherwise pursuant to the Receivership Proceedings, by mutual written consent of the Seller and the Buyer;
- (b) by either Party, upon written notice to the other, if a Governmental Authority issues an order prohibiting the transactions contemplated hereby, which order shall have become final and non-appealable;
- (c) by either Party, if this Agreement is not the Successful Bid (as determined pursuant to the SISP);
- (d) by either Party upon denial of the SISP Order or the Vesting Order (or if any such order is stayed, vacated or varied with the consent of the Parties);
- (e) by the Seller upon written notice to the Buyer if there has been a material violation or breach by the Buyer of any covenant, representation or warranty which would prevent the satisfaction of any condition set forth in Section 6.1 or 6.3 on the Closing Date and such violation or breach has not been waived by the Seller or cured within five (5) days after written notice thereof from the Seller, unless the Seller is in material breach of their obligations under this Agreement; and
- (f) by the Buyer upon written notice to the Seller, if there has been a material violation or breach by the Seller of any covenant (save and except for the covenants set out in Section 7.2), representation or warranty which would prevent the satisfaction of any condition set forth in Section 6.1 or 6.2 on the Closing Date and such violation or breach

has not been waived by the Buyer or cured within five (5) days after written notice thereof from the Buyer, unless the Buyer is in material breach of their obligations under this Agreement.

9.2 Effect of Termination

In the event of termination of this Agreement pursuant to Section 9.1, this Agreement shall forthwith become null and void, except as set forth in Sections 3.2 and Article 11, and nothing herein shall relieve any Party from liability for any breach of this Agreement, or to impair the right of any Party to compel specific performance by any other Party of its obligations under this Agreement.

9.3 Expense Reimbursement and Transaction Fee

- (a) If the Buyer is not the Successful Bidder (as defined in the SISP) pursuant to the SISP, the Buyer shall be entitled to reimbursement for its expenses and a transaction fee in connection with the Transaction in the aggregate amount of \$175,000 the "Expense Reimbursement and Transaction Fee"). Such amount shall be payable by the Successful Bidder to the Buyer in immediately available funds to an account designated by the Buyer concurrently with such closing.
- (b) The payment of the Expense Reimbursement and Transaction Fee shall be approved in the Vesting Order, as contemplated by the SISP, and shall be made in priority to amounts secured by existing security, except as provided in the Vesting Order contemplated by the SISP. Each of the Parties acknowledges that the agreements contained in this Section 9.3 are an integral part of the Transaction and that, without those agreements, the Parties would not enter into this Agreement. The Parties further acknowledge and agree that the Expense Reimbursement and Transaction Fee is a payment of liquidated monetary damages which are a genuine pre-estimate of the costs and damages which the Buyer will suffer or incur as a result of the non-completion of this Agreement, that such payment is not for repayment of current Indebtedness, lost profits or a penalty, and that no Party shall take any position inconsistent with the foregoing. The Seller irrevocably waives any right it may have to raise as a defense that any such liquidated damages are excessive or punitive. Each of the Parties hereby acknowledgesacknowledge and agreesagree that, under circumstances where the Buyer is entitled to the Expense Reimbursement and Transaction Fee and such amounts are paid in full to the Buyer, the Buyer shall be precluded from any other remedy against the Seller at law or in equity or otherwise in respect of damages for non-completion of this Agreement, and in any such case it shall not seek to obtain any recovery, judgment, or damages of any kind, including consequential, indirect, or punitive damages, against the Seller or any of its respective directors, officers, employees, partners, managers, members, shareholders or affiliates in connection with the non-completion of this Agreement or the Transaction.
- (c) Subject to the last sentence of the preceding paragraph, nothing in this Section 9.3 shall preclude any Party from seeking injunctive relief to restrain any breach or threatened breach of the covenants or agreements set forth in this Agreement or otherwise to obtain specific performance of any such covenants or agreement, and any requirement for securing or posting of any bond in connection with the obtaining of any such injunction or specific performance is hereby being waived.

ARTICLE 10 CLOSING

10.1 Location and Time of Closing

The Closing shall take place at the Closing Time on the Closing Date at the Calgary, Alberta offices of Cassels Brock & Blackwell LLP, or at such other location as may be agreed upon by the Parties hereto, which may include by virtual means.

10.2 Closing Deliveries

- (a) At the Closing, the Seller shall deliver to the Buyer the documents required to be delivered by the Seller pursuant to Sections 6.2 and Article 7.
- (b) At the Closing, the Buyer shall deliver to the Seller:
 - (i) the Cash Component, less the Deposit amount;
 - (ii) a duly executed election pursuant to GST Legislation and any certificates, elections or other documents required to be delivered pursuant to Article 7.
 - (iii) the documents required to be delivered by the Buyer pursuant to Section 6.3; and
 - (iv) any other documents reasonably requested by the Seller in order to effect or evidence the consummation of the Transaction or otherwise provided for under this Agreement.

10.3 Receiver's Capacity

The Parties hereby acknowledge and agree that the Receiver is entering into this agreement in its capacity as Court-appointed receiver of the Debtor, and not in its personal capacity.

ARTICLE 11 GENERAL MATTERS

11.1 Confidentiality

- (a) Except to the extent otherwise specifically provided in this Section 11.1, each Party, on behalf of itself and its affiliates, agrees to keep the other Party's Confidential Information confidential and not to use the other Party's Confidential Information in any manner except as required to perform the obligations set out in this Agreement. Each Party agrees to be responsible for any breach of this Section 11.1 by any of its affiliates and its and their respective directors, employees, advisors, agents and representatives.
- (b) Notwithstanding anything to the contrary herein, each Party maintains the right to disclose the other Party's Confidential Information if required to do so by Applicable Laws or requirement of a Governmental Authority, or to appropriate Tax authorities in order to describe the tax treatment and tax structure of the Transaction; provided that the disclosure of such Confidential Information will be limited only to that purpose and provided further that it will use reasonable efforts to cooperate with the other Party in limiting the disclosure of the Confidential Information.
- (c) At the other Party's request, a Party will destroy all of the other Party's Confidential Information, provided that it is permitted to retain one copy of any Confidential Information to the extent required by Applicable Laws or its internal record keeping policies.

(d) Any Confidential Information of the Seller that constitutes part of the Purchased Assets will cease to be Confidential Information of the Seller and will become Confidential Information of the Buyer on Closing.

11.2 Public Notices

No press release or other announcement concerning the Transaction shall be made by the Seller or by the Buyer without the prior consent of the other (such consent not to be unreasonably withheld); provided, however, that subject to the last sentence of this Section 11.2, any Party may, without such consent, make such disclosure if the same is required by Applicable Law (including the Receivership Proceedings) or by any insolvency or other court or securities commission or other similar regulatory authority having jurisdiction over such Party or any of its affiliates, and, if such disclosure is required, the Party making such disclosure shall use commercially reasonable efforts to give prior oral or written notice to the other, and if such prior notice is not possible, to give such notice immediately following the making of such disclosure. Notwithstanding the foregoing: (i) this Agreement may be filed by the Seller with the Court; and (ii) the Transaction may be disclosed by the Seller to the Court, subject to redacting confidential or sensitive information as permitted by Applicable Laws. The Parties further agree that:

- (a) the Seller may prepare and file reports and other documents with the Court containing references to the Transaction and the terms thereof; and
- (b) the Seller and their professional advisors may prepare and file such reports and other documents in the Receivership Proceedings containing references to the Transaction and the terms thereof as may reasonably be necessary to complete the Transaction or to comply with their obligations in connection therewith.

Each of the Parties may issue a press release announcing the execution and delivery of this Agreement, in form and substance mutually agreed to by all of the Parties.

11.3 Survival

The representations and warranties of the Seller in this Agreement or in any agreement, document or certificate delivered pursuant to or in connection with this Agreement or the Transaction are set forth solely for the purpose of Section 6.2 and none of them shall survive Closing. The Seller shall have no liability, whether before or after the Closing, for any breach of the Seller's representations, and the Buyer acknowledges that its exclusive remedy for any such breach shall be termination of this Agreement prior to the Closing (but only if permitted by Section 9.1).

11.4 Non-Recourse

No past, present or future director, officer, employee, incorporator, member, partner, stockholder, affiliate, agent, attorney or representative of the respective Parties hereto, in such capacity, shall have any liability for any obligations or liabilities of the Buyer, the Debtors or the Seller, as applicable, under this Agreement or for any claim based on, in respect of, or by reason of, the Transaction.

11.5 Assignment; Binding Effect

No Party may assign its right or benefits under this Agreement without the consent of the other Party hereto, except that without such consent the Buyer may: (i) assign any or all of its rights and obligations hereunder to one or more of its subsidiaries or affiliates; or (ii) direct that title to all or some of the Purchased Assets be transferred to one or more of its subsidiaries or affiliates, provided that no such assignment or direction shall relieve the Buyer of its obligations hereunder; provided further that if the Buyer shall have assigned all of its rights and obligations hereunder the Buyer shall, immediately following the Closing, be deemed fully released from all of the Buyer's obligations hereunder. References to the Buyer's status under the *Investment Canada Act* are references to the ultimate Buyer. This Agreement

shall be binding upon and inure to the benefit of the Parties and their respective permitted successors and permitted assigns. Nothing in this Agreement shall create or be deemed to create any third Person beneficiary rights in any Person or entity not a Party to this Agreement other than the express third party beneficiaries of Section 11.4 hereof.

11.6 Notices

Any notice, request, demand or other communication required or permitted to be given to a Party pursuant to the provisions of this Agreement will be in writing and will be effective and deemed given under this Agreement on the earliest of: (i) the date of personal delivery; (ii) the date of transmission by email, with confirmed transmission and receipt (if sent during normal business hours of the recipient, if not, then on the next Business Day); (iii) two days after deposit with a nationally-recognized courier or overnight service such as Federal Express; or (iv) five days after mailing via certified mail, return receipt requested. All notices not delivered personally or by facsimile will be sent with postage and other charges prepaid and properly addressed to the Party to be notified at the address set forth for such Party:

(a) If to the Buyer at:

2464525 Alberta Ltd. c/o Tingle Merrett LLP 1250, 639 - 5th Avenue S.W. Calgary, AB T2P 0M9

Attention:Scott ReevesEmail:sreeves@tinglemerrett.com

(b) If to the Seller at:

MNP Ltd. 1500, 640 – 5th Avenue SW Calgary, AB T2P 3G4

Attention: Vanessa Allen Email: vanessa.allen@mnp.ca

with a copy to:

Cassels Brock & Blackwell LLP Bankers Hall West 3810, 888 3rd Street SW Calgary, AB T2P 5C5

Attention:Jeffrey OliverEmail:joliver@cassels.com

Any Party may change its address for service from time to time by notice given in accordance with the foregoing and any subsequent notice shall be sent to such Party at its changed address.

11.7 Counterparts; Facsimile Signatures

This Agreement may be signed in counterparts and each of such counterparts shall constitute an original document and such counterparts, taken together, shall constitute one and the same instrument. Execution of this Agreement by any of the Parties hereto may be evidenced by facsimile, scanned e-mail or internet transmission copy of this Agreement bearing such signature which, for all purposes, shall be deemed to be an original signature.

[Remainder of page intentionally blank]

IN WITNESS WHEREOF the Parties hereto have executed this Agreement as of the date first written above.

MNP LTD., solely in its capacity as receiver of the assets, undertakings and properties of **SOLVAQUA INC.** and not in its personal capacity

By:

Name: Vanessa Allen Title: Senior Vice President

2464525 ALBERTA LTD.

By:

Name: Title:

Schedule 1.1(xx) Purchased Assets

All of the Debtor's right, title and interest in, to and under all of the tangible and intangible assets, properties and rights of every kind and nature and wherever located, other than the Excluded Assets, which relate to or <u>are</u> used or held for use in connection with the Business, including the following:

- (a) the Vivakor Equipment, unless the Buyer elects not to purchase the Vivakor Equipment in the event it is determined by (i) the Receiver, Buyer and Equipment Owner, or (ii) the Court, that the Vivakor Equipment is owned by a Person other than SolvAQUA;
- (b) the SolvAQUA Claims;
- (c) all accounts receivable, notes receivable and negotiable instruments of the Business;
- (d) all prepaid charges and expenses of the Business;
- (e) all inventory, finished goods, raw materials, work in progress, packaging, supplies, parts and other inventories of the Business;
- (f) all equipment and other tangibles assets of the Debtor, including all water treatment equipment, vehicles, tools, parts and supplies, fuel, machinery, furniture, furnishing, appliances, fixtures, office equipment and supplies, owned and licensed computer hardware and related documentation, stored data, communication equipment, trade fixtures and leasehold improvement, in each case, with any transferable warranty and service rights of the Debtor related thereto;
- (g) all Contracts;
- (h) all Intellectual Property;
- (i) all goodwill and intangibles;
- (j) all books and records;
- (k) all rights under insurance contracts and policies;
- (I) all telephone numbers, fax numbers and email addresses;
- (m) all non-disclosure agreements entered into by the Receiver on behalf of the Debtor in connection with the SISP;
- (n) all other additional assets, properties, privileges, rights and interest of the Debtor relating to the Business or the assets of the Debtor of every kind and description and wherever located, whether known or unknown, fixed or unfixed, accrued, absolute, contingent or otherwise, and whether or not specifically referred to in this Agreement; and
- (o) the Assumed Liabilities.

Schedule 1.1(iii) SISP

Schedule 1.1(jjj) Form of SISP Order

Document comparison by Workshare 9.5 on Wednesday, October 19, 2022 1:36:17 PM

Input:	
Document 1 ID	file://C:\Users\dmarechal\Work Folders\CBB System Files\Desktop\VA Final Report 2022 10 11 - DRAFT Stalking Horse APA (no RVO).docx
Description	VA Final Report 2022 10 11 - DRAFT Stalking Horse APA (no RVO)
Document 2 ID	interwovenSite://CASSELS-DMS3/LEGAL/56914030/9
Description	#56914030v9 <legal> - 2022 10 19 - DRAFT Stalking Horse APA (no RVO)</legal>
Rendering set	Standard

Legend:		
Insertion_		
Deletion-		
Moved from-		
Moved to		
Style change		
Format change		
Moved deletion		
Inserted cell		
Deleted cell		
Moved cell		
Split/Merged cell		
Padding cell		

Statistics:			
	Count		
Insertions	24		
Deletions	18		
Moved from	0		
Moved to	0		
Style change	0		
Format changed	0		

Total changes	42
---------------	----